REQUEST FOR AUTHORISATION FOR THE VOLUNTARY AND COMPETING TAKEOVER OFFER FOR SHARES IN APPLUS SERVICES, S.A. MADE BY AMBER EQUITYCO, S.L.U.

This request for authorisation is made public in accordance with article 17 of Royal Decree 1066/2007, of 27 July, on the regime for public offers for acquisition of securities (the "**Royal Decree 1066/2007**") and it relates to an offer that is subject to the mandatory authorisation of the Spanish National Securities Market Commission (Comisión Nacional del Mercado de Valores – the "**CNMV**").

The detailed terms and conditions of the offer will be contained in the prospectus of the offer that will be published after obtaining the mandatory authorisation of the CNMV.

In accordance with article 30.6 of Royal Decree 1362/2007, of 19 October, those shareholders of Applus Services, S.A. acquiring securities attributing voting rights shall notify the CNMV of such acquisition when the proportion of the voting rights held by them reaches or exceeds 1%. In addition, the shareholders already holding 3% of the voting rights in Applus Services, S.A. shall notify any transaction resulting in change in such percentage.

TO THE SPANISH NATIONAL SECURITIES MARKET COMMISSION

Amber EquityCo, S.L.U. (the "**Offeror**"), a Spanish limited liability company whose identification details are set out below, duly represented by Mr Alexander Metelkin, of legal age, of British nationality and holder of valid passport of his nationality number 529231543, and Ms Linda Zhang, of legal age, of German nationality and holder of valid passport of her nationality number C4YMT0ZVV, both duly authorised to act in the name and on behalf of the Offeror by virtue of the decisions made by the joint directors of the Offeror on 13 September 2023,

STATES

1 DECISION TO MAKE THE OFFER

On 13 September 2023, the sole shareholder and the joint directors of the Offeror decided to make a voluntary and competing takeover offer for the acquisition of all the 129,074,133 shares of Applus Services, S.A. ("**Applus**") representing 100% of its share capital (the "**Offer**"), pursuant to the terms and conditions described in this request for authorisation and in the Offer prospectus that has been filed with the CNMV on the date hereof (the "**Prospectus**").

The decision to make the Offer was also taken pursuant to the authorisation passed on 28 June 2023 by the investment committee of ISQ Global Fund III GP, LLC (for and on behalf of the ISQ Fund III, as defined in section 2 below) and on 13 September 2023 by the board of directors of TDR Capital General Partner V Limited (for and on behalf of the TDR Fund V, as defined in section 2 below).

The launching of the Offer does not require the approval of any other resolution by any corporate body of the Offeror or its shareholders.

2 INFORMATION ON THE OFFEROR AND ITS GROUP

The Offeror is Amber EquityCo, S.L.U., a limited liability company (*sociedad de responsabilidad limitada*) incorporated under the laws of Spain, with registered office at Calle Ramírez de Arellano 17, 10^a planta, 28043, Madrid, Spain, registered with the Commercial Register of Madrid under Volume 45136, Page 213, Sheet M-794270, and with Tax Identification Number (N.I.F.) B-13797311. The shares in the Offeror are not admitted to trading on any securities market. Its LEI code is 959800GWS9Z441C74Y15.

The Offeror is wholly and indirectly owned by the English company Amber JVCo Limited ("**Amber JVCo**") through a chain of companies. In particular, the Offeror is wholly owned by the Spanish company Amber BidCo, S.L.U. ("**Amber BidCo**"), which in turn is wholly owned by the English company Amber HoldCo Limited ("**Amber HoldCo**"), which in turn is wholly owned by the English company Amber MidCo 1 Limited ("**Amber MidCo**"), which in turn is wholly owned by the English company Amber JVCo.

Amber JVCo is owned as follows:

- (i) 25% of its share capital is held by Cube Amber USTE HoldCo, LLC, an exempted limited liability company incorporated under the laws of the Cayman Islands.
- (ii) 25% of its share capital is held by Cube Amber UK Holdings Limited (together with Cube Amber USTE HoldCo, LLC, the "ISQ TopCos"), a private limited company incorporated under the laws of England and Wales.

The ISQ TopCos are wholly and indirectly owned by a private equity fund comprised of limited partnerships established in Luxembourg and the Cayman Islands named ISQ Global Infrastructure Fund III ("**ISQ Fund III**"), which is controlled by ISQ Global Fund III GP, LLC ("**ISQ Fund III GP**"), a limited liability company incorporated in the State of Delaware (United States of America). ISQ Fund III GP is wholly owned by ISQ Holdings, LLC ("**ISQ Holdings**"), a Cayman Islands limited liability company which is equally owned and managed by Mr Sadek Wahba, Mr Gautam Bhandari and Mr Adil Rahmathulla, none of whom holds the ability to exercise individual control over ISQ Holdings.

(iii) The remaining 50% of the share capital of Amber JVCo is held by Amber TopCo S.à r.l. ("TDR TopCo"), a limited liability company (société à responsabilité limitée) incorporated under the laws of the Grand Duchy of Luxembourg, which is fully and ultimately owned by TDR Capital V LP ("TDR Fund V"), an English limited partnership controlled by TDR Capital General Partner V Limited ("TDR Fund V GP"), a private limited company incorporated under the laws of Scotland. TDR Fund V GP is wholly and indirectly owned by TDR Capital LLP ("TDR Capital"), an English limited liability partnership with 20 partners, none of whom holds the ability to exercise individual control over TDR Capital.

The Offeror, Amber BidCo, Amber HoldCo, Amber MidCo, Amber JVCo, ISQ TopCos and TDR TopCo are special purpose companies which have been incorporated in order to make the Offer and channel the investment by ISQ Fund III and TDR Fund V in Applus.

A more extensive description of the shareholding and control structure of the Offeror is included in the Prospectus.

3 TYPE OF OFFER

The Offer is of a voluntary nature in accordance with article 13 of Royal Decree 1066/2007 and article 117.1 of Act 6/2023, of 17 March, in Securities Markets and Investment Services (the **"Securities Market Act**").

On 30 June 2023, Manzana Spain BidCo, S.L.U. filed with the CNMV a voluntary takeover offer for the acquisition of all the 129,074,133 shares of Applus at a price in cash of EUR 9.50 per share (the "**Initial Offer**"). The request for authorisation of the Initial Offer was admitted for processing by the CNMV on 17 July 2023.

Therefore, the Offer is a competing offer to the Initial Offer and it is governed by the provisions set forth in Chapter IX of Royal Decree 1066/2007.

The Offer meets the conditions set forth in article 42.1 of Royal Decree 1066/2007 given that it has been filed before the fifth calendar day prior to the end of the Initial Offer's acceptance period, it is addressed to the same number of shares as the Initial Offer and the Offer Price (as defined in section 8 below) is of EUR 9.75, i.e., it is higher than the price of the Initial Offer.

4 OFFEROR'S STAKE IN APPLUS

Neither the Offeror, nor ISQ Holdings or TDR Capital, nor those funds or entities controlled or managed by any of them, nor, to the best of the Offeror's knowledge, any company controlled by the funds ultimately managed or controlled by ISQ Holdings or TDR Capital, or any of the members of their respective governing bodies or their senior management (*personal de alta dirección*), are direct or indirect holders of any Applus shares or act in concert with any third party (person or entity) in connection with the Offer. Therefore, for the purposes of article 5 of Royal Decree 1066/2007, no shares of Applus are attributed to the Offeror.

The Offeror has not appointed any members of the board of directors or the management team of Applus.

In the 12 months prior to the date of this request for authorisation, neither the Offeror, ISQ Holdings or TDR Capital, nor those funds or entities controlled or managed by any of them, nor, to the best of the Offeror's knowledge, any company controlled by funds ultimately managed or controlled by ISQ Holdings or TDR Capital, or any of the members of their respective governing bodies or their senior management (*personal de alta dirección*), have acquired Applus shares or have entered into or agreed to enter into, directly or indirectly, individually or in concert with others, nor carried out in any other manner any transactions involving Applus shares or instruments that could entitle the holder to acquire or subscribe for Applus shares, or that directly or indirectly grant voting rights in Applus.

5 INFORMATION ON APPLUS

Applus Services, S.A. is a public listed company (*sociedad anónima cotizada*) incorporated under the laws of Spain, with registered address at Calle Campezo 1, Edificio 3, Parque Empresarial Las Mercedes, 28022 Madrid, Spain, registered with the Commercial Registry of Madrid under Volume 36874, Page 114, Sheet M-659828 and with Tax Identification Number (N.I.F.) A-64622970. Its commercial name is Applus+.

Applus's share capital amounts to EUR 12,907,413.30, divided into 129,074,133 shares, of EUR 0.10 of face value each, belonging to the same class and series, fully subscribed and paid up. All Applus shares are represented by book entries, whose register is maintained by Sociedad de Gestión de los Sistemas de Registro, Compensación y Liquidación de Valores, S.A. ("**Iberclear**"), and are admitted to trading on the Madrid, Barcelona, Bilbao and Valencia Stock Exchanges through the Stock Exchange Interconnection System (*Sistema de Interconexión Bursátil –SIBE–*) (the "**Spanish Stock Exchanges**").

Applus shares are not admitted to trading on any other regulated market.

According to the information publicly available as of 13 September 2023, Applus holds 146,997 shares in treasury stock, which represent 0.114% of the total share capital of Applus.

According to the information publicly available, there are no subscription rights, or bonds exchangeable or convertible into shares, or warrants, or any other similar instrument that could directly or indirectly entitle its holder to purchase or subscribe for shares of Applus, nor non-voting or special class shares exist.

6 SECURITIES TO WHICH THE OFFER IS ADDRESSED

The Offer is made for the entire share capital of Applus, which is represented by 129,074,133 shares, of EUR 0.10 of face value each, belonging to the same class and series, and fully paid-up and subscribed. The Offer is therefore addressed to all the holders of shares in Applus.

The terms of the Offer, including the Offer Price (as defined in section 8 below), are the same for all the shares in Applus.

7 MARKETS TO WHICH THE OFFER IS ADDRESSED

The Offer is made exclusively in the Spanish market, the only market where Applus' shares are listed. The Offer is addressed to all the shareholders of Applus.

This request for authorisation and its contents do not entail the launching or dissemination of the Offer in any jurisdictions or territories different from the Spanish territory. Therefore, this request for authorisation and the Prospectus, which will be published after the authorisation of the Offer by the CNMV, will not be published, sent to or distributed in any jurisdiction or territory where the publication thereof may be prohibited or restricted by law or where the registration or deposit of additional documentation is required, and any person (including custodians, nominees and trustees) receiving this request for authorisation, the Prospectus or any other documents relating to the Offer shall not directly or indirectly publish or distribute them in such jurisdictions or territories.

The Offer is not being made in or into, and is not capable of acceptance in or from, the United States, and is not being made in or into, and is not capable of acceptance in or from, Canada, Australia, New Zealand, the Republic of South Africa or Japan ("**Other Restricted Jurisdictions**"), and the Prospectus and all other documents relating to the Offer do not constitute or form a part of any offer or solicitation to purchase or subscribe for securities in the United States or any Other Restricted Jurisdiction.

In particular, this request for authorisation will not be published or distributed, neither will the Offer be made, directly or indirectly, in the United States of America, whether by mail or any

other interstate or foreign means or commercial instruments, nor by means of the securities stock exchanges of the United States of America (including its territories and possessions, any state of the United States and the District of Columbia), nor by any other means that may enable the Offer being sent to, or may be distributed in, the United States of America. As mentioned above, this request for authorisation is not a purchase offer neither does it involve an offer to purchase or a solicitation or offer to sell shares in the United States of America.

8 CONSIDERATION OFFERED

The Offer is made as a purchase and sale of shares. The consideration offered by the Offeror to the holders of the shares in Applus amounts to EUR 9.75 per share (the "**Offer Price**"). As a result, the maximum total amount to be paid by the Offeror is EUR 1,258,472,796.75. The Offer Price will be fully paid in cash.

The Offeror has the debt and equity commitments necessary to pay the abovementioned consideration.

If Applus makes or declares any dividend or reserve distribution, equity return or any other kind of distribution to its shareholders, whether of an ordinary or extraordinary nature, interim or complementary, the Offer Price will be reduced in an amount equal to the gross amount per share of the distribution, provided that the date of publication of the result of the Offer in the trading bulletins is the same (or a later date) than the relevant distribution's exdividend date.

The Offeror considers that the Offer Price meets the conditions of an equitable price in accordance with the rules set out in article 9 of Royal Decree 1066/2007 and is not lower than the price resulting from the valuation methodologies of article 10 of Royal Decree 1066/2007.

For this purpose, the Offeror has appointed Kroll Advisory, S.L. as an independent expert to issue a valuation report justifying the Offer Price in accordance with valuation criteria set forth in article 10.5 of Royal Decree 1066/2007 and, consequently, confirm that the Offer Price meets the requirements set out in articles 9 of Royal Decree 1066/2007, for the purpose of its consideration as "equitable price", and 10 of Royal Decree 1066/2007, for the purpose of delisting Applus' shares from the Spanish Stock Exchanges. The valuation report will be filed with the CNMV within the time limit set out in article 17 of Royal Decree 1066/2007.

The Offeror represents that (i) neither the Offeror, ISQ Holdings or TDR Capital, nor those funds or entities controlled or managed by any of them, nor, to the best of the Offeror's knowledge, any company controlled by funds ultimately managed or controlled by ISQ Holdings or TDR Capital, or any of the members of their respective governing bodies or their senior management (*personal de alta dirección*), have acquired or agreed to acquire Applus shares in the 12 months prior to this request for authorisation; (ii) there is no additional consideration that has been or will be paid by the Offeror nor deferred payment for the benefit of any shareholder of Applus; (iii) none of the circumstances of article 9 of Royal Decree 1066/2007 which could trigger a modification of the Offer Price have occurred. Consequently, in the Offeror's opinion, the Offer Price is an equitable price.

In any event, the consideration of the Offer Price as an equitable price is subject to confirmation by the CNMV. To the extent that the CNMV considers that the Offer Price is not an equitable price, the Offeror shall not be obliged to launch a mandatory tender offer if the

Offer is accepted by at least 50% of the voting rights to which the Offer is addressed, excluding from the calculation those already owned by the Offeror and those corresponding to shareholders who have reached any agreement with the Offeror in relation to the Offer.

Notwithstanding that the amounts of the premiums on the trading prices provided for below may change after the date of this request for authorisation depending on the trading prices, and that these amounts do not entail that the price may be considered equitable in terms of article 9 of Royal Decree 1066/2007, the Offeror indicates that the Offer Price represents a premium of:

- 4.95% over the trading price of the Applus shares as at market close on 28 June 2023, the trading day immediately prior to the filing of the request for authorisation of the Initial Offer, which was EUR 9.29 per Applus share;
- (ii) 7.85% over the volume-weighted average trading price of the Applus shares during the month immediately prior to the filing of the request for authorisation of the Initial Offer, which was EUR 9.04 per Applus share;
- (iii) 16.63% over the volume-weighted average trading price of the Applus shares during the quarter immediately prior to the filing of the request for authorisation of the Initial Offer, which was EUR 8.36 per Applus share;
- (iv) 28.12% over the volume-weighted average trading price of the Applus shares during the six-months immediately prior to the filing of the request for authorisation of the Initial Offer, which is EUR 7.61 per Applus share;
- 40.49% over the volume-weighted average trading price of the Applus shares during the year immediately prior to the filing of the request for authorisation of the Initial Offer, which is EUR 6.94 per Applus share; and
- (vi) 2.63% over the price of the Initial Offer, which is EUR 9.50 per Applus share.

9 GUARANTEE FOR THE OFFER

In accordance with article 15 of Royal Decree 1066/2007, the Offeror will file with the CNMV first demand bank guarantees for an aggregate amount of EUR 1,258,472,796.75 (i.e., the maximum total amount to be paid by the Offeror pursuant to the Offer).

10 PRIOR AUTHORISATION UNDER ARTICLE 26.2 OF ROYAL DECREE 1066/2007

Pursuant to article 7 bis of Act 19/2003, of 4 July, on the legal regime governing capital movements and economic transactions abroad and on certain measures to prevent money laundering, and the Single Transitory Provision of Royal Decree-law 34/2020, of 17 November, on urgent measures supporting the business solvency and the energy sector, and on tax matters, the acquisition by the Offeror, and indirectly by ISQ Fund III and TDR Fund V, of a stake in Applus as a result of the Offer will be subject to the prior authorisation from the Council of Ministers of the Government of Spain.

The Offeror will submit the application for the relevant authorisation to the Directorate General for International Trade and Investment of the Ministry of Industry, Trade and Tourism as soon as possible and in cooperation with such authority.

In accordance with article 26.2 of Royal Decree 1066/2007, the CNMV will not approve the Offer until it receives evidence that the approval of the Council of Ministers of the Government of Spain set forth in this section 10 has been obtained.

In the event that the authorisation from the Council of Ministers of the Government of Spain is subject to any condition or obligation imposed on the Offeror, the Offeror will notify the CNMV and the market as soon as possible whether or not it accepts them. If the Offeror communicates its decision to not accept any of the aforementioned conditions or obligations, the Offeror will deem that the relevant authorisation has not been obtained and, consequently, it will proceed to withdraw the Offer.

11 CONDITIONS FOR THE EFFECTIVENESS OF THE OFFER

In accordance with articles 13 and 26 of Royal Decree 1066/2007, the effectiveness of the Offer will be subject to the satisfaction of the following conditions:

- (i) Pursuant to article 13.2b) of Royal Decree 1066/2007, the acceptance of the Offer by at least 96,805,600 Applus shares, representing 75% of Applus' voting share capital.
- (ii) Pursuant to article 26.1 of Royal Decree 1066/2007, the Offeror obtaining the approvals, with no conditions, from the merger control authorities indicated in sections 12.1(i) to 12.1(xi), both inclusive.
- (iii) Pursuant to article 13.2d) of Royal Decree 1066/2007, the Offeror obtaining the approvals, with no conditions, from the regulatory authorities indicated in sections 12.2 and 12.3 below.
- (iv) Pursuant to article 13.2d) of Royal Decree 1066/2007, the Offeror obtaining the express waiver, with no conditions, from (a) the Catalan Government (*Generalidad de Cataluña*) in relation to IDIADA Automotive Technology, S.A. and LGAI Technological Center, S.A.; and (b) the Irish Road Safety Authority in relation to Applus Inspection Services Ireland Limited, in both cases to their respective rights that they could exercise as a result of the indirect change of control in the share capital of the aforementioned Applus' subsidiaries which would occur as a consequence of the settlement of the Offer.

12 NECESSARY ANTITRUST AND OTHER REGULATORY APPROVALS

12.1 Merger control authorisations

According to the information available, the Offeror considers that the economic concentration resulting from the Offer requires authorisation from the following merger control authorities:

- authorisation (or deemed authorisation) by the European Commission, given that the acquisition of control of Applus arising from the Offer meet the applicable thresholds under Regulation (EC) 139/2004, of 20 January 2004, on the control of concentrations between undertakings;
- (ii) expiration or early termination of the applicable waiting periods pursuant to the United States of America's Hart-Scott-Rodino Antitrust Improvements Act 1976;
- (iii) authorisation (or deemed authorisation) by the National Antitrust Authority of Canada (BCA) pursuant to the RSC Antitrust Defence Act 1985;

- (iv) authorisation (or deemed authorisation) by the General Authority for Competition under the provisions of the Competition Law of 6 March 2019 (Royal Decree No M75) of the Kingdom of Saudi Arabia;
- (v) authorisation (or deemed authorisation) by the State Administration for Market Regulation (SAMR) pursuant to the Anti-Monopoly Law of the People's Republic of China;
- (vi) authorisation (or deemed authorisation), or acknowledgment of receipt (in case of short-from notification), as applicable, of the Colombian Superintendence of Industry and Commerce ("SIC") pursuant to article 9 of Law 1340/2009;
- (vii) authorisation (or deemed authorisation) by the Chilean competition enforcement agency (*Fiscalía Nacional Económica – FNE*) under Title IV, on concentration transactions, of Decree Law 211 of 1973, as amended (*Decreto Ley Nº 211, de 1973, y sus modificaciones - "DL 211"*), or otherwise by the Chilean competition court (*Tribunal de Defensa de la Libre Competencia - TDLC*), or if applicable, by the Supreme Court of Chile, pursuant to numeral 5) of Article 18, Article 31 bis and the final paragraph of Article 57, of DL 211;
- (viii) authorisation (or deemed authorisation) by the Administrative Council of Economic Defence of Brazil (*Conselho Administrativo de Defesa Econômica CADE*) pursuant to the Brazilian Competition Act (*Law nº 12,529 of November 30, 2011*);
- (ix) authorisation (or deemed authorisation) by the Angolan Competition Authority ("Autoridade Reguladora da Concorrência" – "ARC") pursuant to the Angolan Competition Law (Law 5/18 of 10 May) and the Angolan Regulation on the Competition Law (Presidential Decree 240/18 of 12 October);
- authorisation (or deemed authorisation) by the Federal Competition and Consumer Protection Commission (FCCPC) of Nigeria pursuant to the Federal Competition and Consumer Protection Act 2018;
- (xi) authorisation (or deemed authorisation) by the Kuwait Antitrust Protection Agency under the provisions of the Antitrust Defence Law 72/2020;
- (xii) authorisation (or deemed authorisation) of the Indonesian Competition Commission (*Komisi Pengawas Persaingan Usaha – "KPPU*") pursuant to Regulation No. 3 of 2023 on Merger Filing Procedures; and
- (xiii) authorisation (or deemed authorisation) of the Antitrust Authority, Secretary of Trade or Argentine Antitrust Commission (*Comisión Nacional de Defensa de la Competencia - "CNDC"*) pursuant to the Antitrust Law (Law No. 27,442 of May 15, 2018).

The Offeror will proceed to carry out the filings and notifications to the relevant merger control authorities as soon as possible and in cooperation with such authorities.

12.2 Foreign investment authorisations in countries other than Spain

It is expected that the Offer may require the authorisation from the relevant authorities under the foreign investment regulations applicable in certain jurisdictions other than Spain where the Applus group operates ("**Other FDI Acts**").

Once the necessary information on Applus' activities in the different jurisdictions has been obtained and the relevant analysis has been completed, the Offeror will inform the CNMV

(for its publication by way of OIR –"other relevant information" or "otra información relevante" in Spanish–) of the authorisations that the Offer will require pursuant to such Other FDI Acts and which the Offeror decides to maintain as conditions for the Offer considering their materiality for the business of the Applus Group and their relevance to the interests of ISQ Holdings, TDR Capital and the companies controlled by funds ultimately managed or controlled by ISQ Holdings or TDR Capital.

The Offeror will initiate as soon as practicable the procedure to obtain any such authorisations before the relevant authorities and in cooperation with them.

12.3 Authorisation from the European Commission under the FSR Regulation

According to the information available, it is expected that the Offer may require authorisation from the European Commission in accordance with Regulation (EU) 2022/2560 of the European Parliament and of the Council, of 14 December 2022, on foreign subsidies distorting the internal market (the "**FSR Regulation**").

Once the necessary information is gathered and the corresponding analysis is completed, the Offeror will submit the relevant application form to the European Commission as soon as possible and in cooperation with such authority.

12.4 Non-objection from the Luxembourg Insurance Commission (*Commissariat aux Assurances*)

In accordance with articles 87 and et seq. of the Luxembourg Law of 7 December 2015 on the insurance sector (*loi du 7 Decembre 2015 sur le secteur des assurances*), the acquisition by the Offeror and its group of an indirect interest of, at least, 10% in the share capital of Libertytown RE SA, a captive reinsurance company authorised in Luxembourg, which is a wholly-owned subsidiary of Applus, is subject to the Offeror and its group obtaining a non-objection decision from the Luxembourg Insurance Commission (*Commissariat aux Assurances*), either express or by absence of a decision within the legal deadline.

The Offeror will file the relevant non-opposition request as soon as possible and in cooperation with such authority.

13 AGREEMENTS RELATING TO THE OFFER AND APPLUS

13.1 Confidentiality agreement

On 25 April 2023, Applus, I Squared Capital Advisors (UK) LLP and TDR Capital, entered into a confidentiality agreement to preserve the use and confidentiality of the initial discussions and the information that Applus might make available to both of them to analyse the viability of the Offer.

13.2 Joint bid agreement

On 13 September 2023, certain limited partnerships belonging to ISQ Fund III (the "**ISQ Main Funds**"), on the one hand, and TDR Fund V, on the other hand, entered into a joint bid agreement (the "**Joint Bid Agreement**"), pursuant to which they agreed certain cooperation principles for the implementation of the Offer. Under the terms of the Joint Bid Agreement, the ISQ Main Funds and TDR Fund V have undertaken to work with each other on an exclusive basis to implement the Offer and have agreed to adopt all material decisions with respect to the conduct of the Offer unanimously.

Pursuant to the Joint Bid Agreement, the ISQ Main Funds and TDR Fund V will enter into a shareholders' agreement relating to Amber JVCo which will set out the terms appended to the Joint Bid Agreement (the "**Term Sheet**").

Pursuant to the terms of the Term Sheet, subject to certain step-down provisions, each of ISQ Main Funds and TDR Fund V will be entitled to appoint and remove four directors to the board of directors of Amber JVCo.

Likewise, and subject to customary permitted transfer exceptions, ISQ Main Funds and TDR Fund V may not dispose of their equity stake in Applus without the consent of the other until the date falling two years from settlement of the Offer. After this period, ISQ Main Funds or TDR Fund V may dispose of their interests subject to certain rights of the non-transferring shareholder.

A copy of the Joint Bid Agreement and the Term Sheet is attached as an <u>Annex</u> to this request for authorisation.

Other than the agreements described in this section, there is no agreement of any nature in relation to the Offer or to Applus entered into between, on the one hand, the Offeror or any of the entities named in section 2 above, and, on the other hand, Applus, any of its shareholders or the members of the management, executive and controlling bodies of Applus, nor have any advantages been reserved for the shareholders of Applus or for the members of such bodies of Applus.

14 INITIATIVES ON TRADING MATTERS

If the requirements set out in article 116 of the Securities Market Act and article 47 of Royal Decree 1066/2007 are met, the Offeror intends to exercise its squeeze-out right with respect to the remaining shares in Applus at the Offer Price (subject to any adjustment in accordance with section 8 in the event of a distribution of dividends or other distributions to Applus shareholders). The execution of the squeeze-out transaction will give rise, in accordance with articles 47 and 48 of Royal Decree 1066/2007 and related provisions, to the delisting of Applus' shares from the Spanish Stock Exchanges.

If the requirements for the squeeze-out are not met but the Offeror reaches on the date of the settlement of the Offer a minimum shareholding of 75% of Applus' voting rights, the Offeror intends to delist Applus' shares from the Spanish Stock Exchanges in accordance with the delisting procedure with exception of delisting offer provided in article 11.d) of Royal Decree 1066/2007, and to that effect the Offeror will provide the valuation report referred to in section 8 above justifying that the Offer Price complies with the valuation criteria set out in article 10 of Royal Decree 1066/2007.

If the Offeror does not reach 75% of the voting rights in Applus on the settlement date of the Offer, the Offeror will consider waiving the minimum acceptance condition and, subsequently, promoting the delisting of Applus' shares from the Spanish Stock Exchanges, which would require a subsequent delisting offer in accordance with article 65 of the Spanish Securities Market Act and article 10 of Spanish Royal Decree 1066/2007.

15 OTHER INFORMATION

In the opinion of the Offeror, as of the date of this request for authorisation there is no other information which may be necessary for a good understanding of the Offer, other than the information included in this request for authorisation

16 DOCUMENTS ATTACHED TO THIS REQUEST FOR AUTHORISATION

In accordance with article 17 of Royal Decree 1066/2007 and Annex II of Circular 8/2008 of the CNMV, the following documents are attached to this request for authorisation:

- (i) a duly signed copy of the Prospectus;
- (ii) documentation evidencing the decision to make the Offer approved by the joint directors and the sole shareholder of the Offeror;
- (iii) documentation evidencing the authorisation in relation to the Offer passed by the investment committee of ISQ Global Fund III GP, LLC;
- (iv) documentation evidencing the authorisation in relation to the Offer passed by the board of directors of TDR Capital General Partner V Limited;
- (v) certificate evidencing the incorporation of the Offeror and its current articles of association, issued by the Commercial Registry of Madrid;
- (vi) legalised and apostilled copy of the limited liability company agreement of ISQ Global Fund III GP, LLC, as well as of the certificate of formation issued by the Division of Corporations of the Secretary of State of Delaware, evidencing its incorporation and registration, both together with the respective sworn translations into Spanish;
- (vii) legalised and apostilled copy of the articles of association of TDR Capital General Partner V Limited, as well as of the extract (excerpt) issued by the Companies House of the United Kingdom, evidencing its incorporation and registration, both together with the respective sworn translations into Spanish;
- (viii) Joint Bid Agreement and Term Sheet;
- (ix) certificate of the individual unaudited financial information of the Offeror as of 1 August 2023;
- (x) copy of ISQ Global Infrastructure Fund III 2022 combined annual accounts audited by PriceWaterhouseCoopers; and
- (xi) Offeror's letter regarding the publicity of the Offer.

Pursuant to article 20 of Royal Decree 1066/2007, the remaining necessary documents will be submitted to the CNMV within seven business days following the date of filing of this request for authorisation.

17 NOTIFICATIONS

Below is the Offeror's address for notifications and communications in relation to this request for authorisation and the relevant file with the CNMV:

Linklaters, S.L.P.

Attn:	Esteban Arza
Address:	Calle Almagro 40, 28010 Madrid
Tel.:	+34 913996100
Fax:	+34 913996101
E-mail:	esteban.arza@linklaters.com

In light of the above, the Offeror

REQUESTS

The Spanish National Securities Market Commission to consider this request for authorisation, together with the Prospectus of the Offer and other documents attached and admit them for processing and to authorise the launching of the Offer.

In accordance with Annex II of Circular 8/2008 of the CNMV, it is noted that this request for authorisation is deemed to be the announcement of the Offer for all purposes.

Madrid, 14 September 2023.

Amber EquityCo, S.L.U.

Mr Alexander Metelkin

Amber EquityCo, S.L.U.

Ms Linda Zhang

English version for information purposes only. Spanish version prevails.

Annex Joint Bid Agreement and Term Sheet

Linklaters

Joint Bid Agreement

Dated 13 September 2023

TDR FUND

and

ISQ FUNDS

Contents Pag	
1	Definitions and Interpretation1
2	Conduct of the Offer
3	Regulatory Filings9
4	Bid Financing10
5	Standstill11
6	Exclusivity and Syndication 12
7	Warranties, Undertakings and Acknowledgements 14
8	Indemnification 15
9	Withdrawal and Termination
10	Transaction Costs
11	Engagement and Reports and Reliance 17
12	Confidentiality and Announcements
13	Assignment and Other Dealings 19
14	Amendments and Waivers 19
15	Severability19
16	Counterparts
17	Third party rights
18	Governing Law and Jurisdiction

Table of Contents

This Joint Bid Agreement (the "Agreement") is entered into on 13 September 2023 between:

- (1) The entity whose name and address is set out in Part A of Schedule 1, ultimately controlled by TDR Capital General Partner V Limited, acting in its capacity as general partner of TDR Capital General Partner V L.P. (the "TDR Fund"); and
- (2) Those entities whose names and addresses are set out in Part B of Schedule 1 (collectively the "ISQ Funds" and each an "ISQ Fund"),

(together, the "**Parties**" and the TDR Fund on the one hand and each of the ISQ Funds on the other hand a "**Party**").

Whereas:

- (A) The Parties intend to explore the potential voluntary public tender offer (the "Offer") for the acquisition of the entire issued share capital of Target for cash consideration in accordance with the Royal Decree 1066/2007, of July 27, on the rules for public tender offers (*Real Decreto 1066/2007, de 27 de julio, sobre el régimen de las ofertas públicas de adquisición de valores*) (the "Tender Offer Regulations") and the future management of Target by Amber JVCo Limited, a company jointly owned and operated (directly or indirectly) by the Parties ("Amber JVCo") following completion of the Offer (the "Transaction").
- (B) It is intended that the Offer will be implemented by Amber EquityCo, S.L.U., an indirect wholly-owned subsidiary of Amber JVCo ("EquityCo").
- (C) The Parties wish to evaluate whether to implement the Offer. This Agreement sets out the terms and conditions of an agreement between the Parties in connection with the conduct of such evaluation and, if relevant, the implementation of the Offer, the terms and conditions of which will be contained in the Offer Documentation, pursuant to and in accordance with the Tender Offer Regulations.

It is agreed as follows:

1 Definitions and Interpretation

1.1 Definitions

In this Agreement the following words and expressions shall have the following meanings:

"Acting in Concert" has the meaning given in article 5 of the Tender Offer Regulations;

"Advisers" means those advisers as agreed between the Parties from time to time;

"Affiliate" means, in respect of a Party, any person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with that Party from time to time and includes any funds and/or vehicles managed and/or advised by that Party or its Affiliates within the meaning of the foregoing but excludes: (a) any portfolio or investee companies in which any such funds and/or vehicles directly or indirectly hold an interest or investment; (b) such funds or vehicles which engage primarily in investment in debt and/or debt securities; and (c) any other person that is not involved, directly or indirectly, in the private equity business of that person and has not received information in respect of the Offer;

"Agreement" has the meaning given in the recitals;

"Amber JVCo" has the meaning given in the recitals;

"**Amber JVCo Group**" means Amber JVCo together with its subsidiary undertakings from time to time;

"BidCo" means Amber BidCo, S.L.U.;

"**Budget**" means the budget for the Joint Bid Costs, ISQ Costs and TDR Costs as may be agreed between the Parties from time to time;

"CNMV" means the Spanish Securities Market Commission (Comisión Nacional del Mercado de Valores);

"**Companies Act Spain**" means Royal Legislative Decree 1/2010, of 2 July, approving the consolidated text of the companies act;

"**Companies Act UK**" means the Companies Act 2006 as amended at any time prior to the date hereof;

"**Concert Parties**" means, in respect of a Party, collectively, any persons Acting in Concert with such Party, provided that, for the purpose of this Agreement only, neither Party or its Affiliates shall be deemed to be Concert Parties of the other Party or its Affiliates;

"Confidential Information" means:

- (a) all information (in whatever form) supplied by or on behalf of either Party, any of its Affiliates or any of their respective Representatives, whether before, on or after the date of this Agreement, in connection with the Offer, including any analyses, reports or documents which contain or reflect, or are derived or generated from, any such information;
- (b) any information supplied by Target or its professional advisers at any time after any approach by a Party to the board of directors of Target in connection with the Offer;
- (c) this Agreement and any other transaction documents in relation to the Offer; and
- (d) the existence, status or progress or any negotiations or discussions relating to the Offer;

"**Control**" with respect to a person: (a) ownership of more than 50 per cent. of the voting securities of such person; (b) the right to appoint, or cause the appointment of, more than 50 per cent. of the members of the board of directors (or similar governing body) of such person; or (c) the right to manage, or direct the management of, on a discretionary basis, the business, affairs and/or assets of such person, and a general partner of a limited partnership is deemed to Control such limited partnership and a permanent investment manager of a fund is deemed to Control such fund (and the terms "**Controlling**" and "**Controlled**" shall have meanings correlative to the foregoing);

"Criminal Code Spain" means Organic Act 10/1995, of 23 November, on the criminal code;

"Defaulting Party" has the meaning given in Clause 4.4;

"ECL" has the meaning given in Clause 4.1;

"Effective Date" means the date upon which the Offer is settled in accordance with article 37 of the Tender Offer Regulations;

"**Engagement Letters**" means each and every engagement letter provided by any Adviser in connection with the Transaction for the benefit, either directly or indirectly, of the Parties (but excluding any engagement letter for the sole benefit of only one of the Parties); "Equity Commitment" has the meaning given in Clause 2.7;

"EquityCo" has the meaning given in the recitals;

"EquityCo Board" means the board of directors of EquityCo from time to time;

"ERISA" shall mean the United States Employee Retirement Income Security Act of 1974, as amended;

"Filing Date" means the earlier of: (a) the date on which the first version of the Offer Prospectus is filed with the CNMV; and (b) if applicable, the initial announcement of the Offer in accordance with article 16 of the Tender Offer Regulations;

"Financing" has the meaning given in Clause 2.3.8;

"FinCo" means Amber FinCo PLC;

"Funds" means the TDR Fund or the ISQ Funds (as applicable);

"Indemnification Agreement" means each and every agreement in which any Party or its Affiliates (excluding members of the Amber JVCo Group) agrees to indemnification or other obligations in connection with the Transaction in respect of, either directly or indirectly, the Parties (but excluding any agreements entered into for the sole benefit of only one of the Parties);

"Initial Commitments" has the meaning given in Clause 4.1;

"Initial Investors" has the meaning given in Clause 4.1;

"Initial Investments" has the meaning given in Clause 4.1;

"Investment Date" has the meaning given in Clause 4.1;

"Investors" has the meaning given in Clause 4.1;

"**ISQ**" means ISQ Global Fund III GP, LLC, a Delaware limited liability company with its registered address at 251 Little Falls Drive, Wilmington, DE 19808;

"**ISQ Costs**" means all costs, fees and expenses incurred in relation to advice specific only to ISQ and its Affiliates;

"**JVCo**" means Amber JVCo, EquityCo and any other entities incorporated by the Parties in relation to the Offer, the parent entity of which shall be Amber JVCo;

"Joint Bid Costs" means, subject to Clause 10.6, costs, fees and expenses incurred by (or on behalf of) any JVCo or by either Party for the benefit of both Parties, including TDR Costs and ISQ Costs;

"Linklaters" means Linklaters LLP and Linklaters, S.L.P.;

"Losses" means any losses, claims, damages or liabilities (and any reasonable and documented professional fees and out-of-pocket expenses in connection therewith);

"Non-Defaulting Party" has the meaning given in Clause 4.4;

"Offer" has the meaning given in the recitals;

"Offer Committee" has the meaning given in Clause 2.9;

"Offer Committee Members" has the meaning given in Clause 2.9;

"Offer Documentation" has the meaning given in Clause 2.3.6;

"Offer Period" means the period from the Filing Date until the earliest of (i) the Effective Date, or (ii) the date on which (a) the Parties agree, subject to Tender Offer Regulations, on the withdrawal of the Offer, or (b) the Offer conditions are neither satisfied nor waived on time rendering the Offer ineffective;

"Offer Prospectus" means the prospectus setting out the terms and conditions of the Offer;

"**Overlapping Syndicatee**" shall mean any potential syndicatee designated as an Overlapping Syndicatee by Kirkland & Ellis International LLP from time to time by virtue of such potential syndicatee being an existing limited partner or investor in both the TDR Fund and an ISQ Fund;

"Parties" has the meaning given in the recitals;

"Payee" has the meaning given in Clause 8;

"Payor" has the meaning given in Clause 8;

"Permitted Syndicatee" has the meaning given in Clause 6.3;

"Permitted Syndication" has the meaning given in Clause 6.3;

"**Potential Syndicatee**" means (i) any Overlapping Syndicatee that is agreed by the Parties to be a Potential Syndicatee of a Party and (ii) any person that is not an Overlapping Syndicatee or Restricted Syndicatee;

"**Relevant Date**" means, in respect of any Party, the date that is six months after the earlier of: (i) the date on which such Party withdraws from the Offer in accordance with Clause 9.1 below; and (ii) the date on which this Agreement terminates or is terminated in accordance with Clause 9.3;

"**Relevant Proportion**" means, in respect of a Party, the proportion that the Initial Commitment of the Initial Investor(s) of such Party bears to the aggregate of all Initial Commitments being 50 per cent. each;

"Remaining Party" has the meaning given in Clause 9.2.2;

"**Reports and Reliance Letters**" means each and every due diligence report (and associated reliance letter or terms) in connection therewith provided by any Adviser in connection with the Transaction for the benefit of the Parties (but excluding any due diligence reports for the sole benefit of only one of the Parties);

"**Representatives**" means, in respect of any person, its partners, officers, employees professional advisers, lenders, proposed lenders, auditors and other representatives of such person;

"Restricted Syndicatee" means:

(a) any person who (as a result of such syndication) would: (i) cause the Amber JVCo Group to become subject to any regulatory regime that would impose adverse restrictions or obligations on the Amber JVCo Group; (ii) be required to be registered, licensed, approved or authorised by any regulator; (iii) be required (or require EquityCo or any Party) to make an anti-trust, foreign investment or regulatory filing not contemplated by the Offer Prospectus filed with the CNMV; (iv) be reasonably likely to cause a material delay in, or jeopardise receipt of, any approval or clearance that is required to acquire or operate the Amber JVCo Group or undertake any transaction or arrangement approved by the EquityCo Board or the board of any other JVCo; or (v) otherwise be reasonably likely to have any materially adverse effect on the regulatory or legal position of the Amber JVCo Group;

- (b) any person who is subject to insolvency proceedings or analogous events;
- (c) any person who is not (prior to such syndication) an existing limited partner or investor in (i) the TDR Fund or any of the ISQ Funds or (ii) any fund and/or vehicle under common management or control with the TDR Fund or any of the ISQ Funds; or
- (d) any person that is an existing shareholder of Target.

"Securities Market Act Spain" means Act 6/2023, of 17 March, on Securities Markets and Investment Services;

"Shareholders' Agreement" has the meaning given in Clause 2.3.6(i);

"**Structure Paper**" means the structure paper relating to the Transaction prepared by Linklaters dated 17 August 2023;

"syndication" means any syndication, transfer or assignment (and "syndicate", "transfer" and "assign" and related words and expressions shall have meanings correlative to the foregoing);

"subsidiary" has the meaning given to such term in the Companies Act UK;

"Syndication Interests" has the meaning given in Clause 6.3;

"Target" means Amber Services, S.A.;

"Target Group" means Target and its subsidiaries;

"**Target Securities Interest**" means equity and debt securities issued by the Target Group (including, among others, warrants or other financial instruments carrying the option to acquire or subscribe for shares);

"TDR" means TDR Capital General Partner V Limited, a limited company incorporated in Scotland with registered number SC707592, whose registered office is at 50 Lothian Road, Festival Square, Edinburgh, Scotland, EH3 9WJ and any references herein to TDR shall be to TDR acting in its capacity as general partner of the TDR Fund;

"**TDR Costs**" means all costs, fees and expenses incurred in relation to advice specific only to TDR and its Affiliates;

"Tender Offer Regulations" has the meaning given in the recitals;

"**Term Sheet**" means the term sheet agreed between the Parties which sets out the principal terms on which the Parties propose to invest in Amber JVCo in connection with the Transaction, a copy of which is set out at Schedule 2 of this Agreement; and

"Transaction" has the meaning given in the recitals.

1.2 Interpretation

1.2.1 Clause headings shall not affect the interpretation of this Agreement.

- **1.2.2** Unless the context otherwise requires, words in the singular shall include the plural and in the plural shall include the singular.
- **1.2.3** A reference to a statute or statutory provision is a reference to it as amended, extended or re-enacted from time to time and shall include all subordinate legislation made from time to time under that statute or statutory provision.
- **1.2.4** References to Clauses are to the clauses of this Agreement and references to articles are to the relevant article of the Tender Offer Regulations.
- **1.2.5** The words "including", "include", "in particular" and words of similar effect shall not be deemed to limit the general effect of the words that precede them.

2 Conduct of the Offer

- **2.1** The Parties shall agree and implement a strategy for making the Offer.
- 2.2 The Parties shall, prior to the Filing Date, incorporate the JVCos which are required prior to the Filing Date for the purpose of the proposed Offer and shall as required agree in good faith any further matters in relation to the JVCos, including their ownership structure, to obtain a mutually acceptable and beneficial structure for the Offer, and for holding any shares in Target acquired by EquityCo pursuant to the Offer, in each case as contemplated by the Structure Paper. All JVCos are and shall continue to be until the Effective Date, unless otherwise agreed in writing by the Parties, owned (directly or indirectly) by the Parties in the Relevant Proportions.
- **2.3** Subject always to the provisions of Clause 8, the Parties agree to work together in good faith towards reaching unanimous agreement on:
 - **2.3.1** the pricing and other offer terms including the manner of announcement and implementation of the Offer (and any pricing strategy or revisions relating thereto);
 - **2.3.2** the manner and timing of all discussions with Target, its management and any of its shareholders or other stakeholders;
 - **2.3.3** the general conduct of the Offer;
 - 2.3.4 the structure of the Offer, including finalising the Structure Paper (it being acknowledged that a Party will take all reasonable steps to accommodate the other Party's structuring requirements in respect of ERISA);
 - 2.3.5 the incorporation of any JVCo and the corporate governance arrangements of such entities, including the board composition of any such entity, in particular appointing one or more persons from each Party to facilitate and take responsibility under the Tender Offer Regulations (together with such other persons as may take responsibility under Tender Offer Regulations) for each Party as may be reflected in the Offer Prospectus;
 - **2.3.6** the definitive documentation required to implement the Offer (the "**Offer Documentation**"), including:
 - a shareholders' agreement relating to Amber JVCo (the "Shareholders' Agreement") and articles of association or other constitutional documents of the JVCos, the terms of which shall be consistent with the Term Sheet;

- the public documentation necessary or desirable in connection with any announcement relating to the Offer and implementation the Offer, including the Offer Prospectus;
- (iii) the form of irrevocable undertakings that may be given in connection with the Offer;
- (iv) the ECL and such other documents required for certain funds financing purposes;
- (v) the documents required for the Financing (as defined below);
- (vi) the terms of appointment of any adviser as set out in an Engagement Letter between the relevant adviser and any JVCo and/or either Party; and
- (vii) any other agreements as may be determined necessary or desirable in connection with the announcement and implementation of the Offer;
- 2.3.7 the strategy and financing of any market purchases of Target Securities Interests;
- **2.3.8** the debt financing of the Offer (the "**Financing**"), including the selection and appointment of financing banks, arrangers and other advisers (other than the financing banks set out in Schedule 2);
- 2.3.9 the strategy for the syndication of the Financing;
- 2.3.10 the appointment (other than the advisers set out in Schedule 2) or instruction of any advisers to or on behalf of any JVCo (but, for the avoidance of doubt, this shall not apply to any appointment or instruction given to advisers acting on behalf of any Party where the work undertaken pursuant to such appointment or instruction does not give rise to the incurrence of Joint Bid Costs);
- 2.3.11 the scope of confirmatory due diligence on Target;
- **2.3.12** any decision relating to Target's management;
- **2.3.13** any decision regarding the seeking or making of an application to cancel the admission to trading of Target;
- **2.3.14** the actual or purported waiver, treating as satisfied, invocation or amendment of any condition or of any pre-condition to the Offer, the extension of any acceptance period of the Offer or similar and/or the revision of terms of the Offer, or the lapsing or withdrawal of the Offer;
- 2.3.15 the strategy to delist Target;
- 2.3.16 any decision regarding any remedies or divestitures that might be required by a regulatory or governmental authority in connection with the Offer concerning Target, including all matters regarding the negotiations and terms and conditions of any such remedies or divestitures. For the avoidance of doubt, where such decision solely concerns the asset(s) of any Party (and not Target), such decision shall be made at the sole discretion of that Party (and if such decision concerns an asset in which both Parties are invested, the decision shall be made only with the consent of both Parties);

- **2.3.17** any decision to waive any regulatory condition in respect of the Transaction or any decision to proceed with completion of the Transaction notwithstanding that any such regulatory condition has not been satisfied; and
- **2.3.18** subject always to the requirements of the Tender Offer Regulations or other applicable law or regulation to which the Parties are directly or indirectly subject and Clause 2.7, the timing of the release of the public documentation contemplated by Clause 2.3.6(ii) or any other announcement in connection with the Offer.
- **2.4** For the avoidance of doubt, the Parties shall not file the Offer Prospectus with the CNMV or otherwise make any announcement of the Offer until they have reached unanimous agreement on each matter set out in Clause 2.3 above to the extent that such decisions are necessary to be taken prior to the Filing Date.
- **2.5** Neither Party shall, and each Party shall procure that none of its Concert Parties (including, so far as it is within its powers, any JVCo) shall:
 - **2.5.1** file the Offer Prospectus with the CNMV until the Parties have agreed in writing to the timing, form and content of such filing;
 - 2.5.2 subject to Clause 2.6 below, make any public announcement or external communication in connection with the Offer, whether formal or informal, until the Parties have consented in writing to the timing, form and content of such announcement or communication (such consent not to be unreasonably withheld or delayed); or
 - 2.5.3 save to the extent required by law, or any securities exchange or regulatory or governmental body to which the Party or any Affiliate is subject (including the CNMV), liaise, negotiate, or otherwise communicate with Target or its shareholders or advisers, any regulatory authority or exchange, government body, including any rating agencies or the CNMV with respect to the Offer,

in all cases without the prior consent of the other Party. Notwithstanding the foregoing, Linklaters shall be entitled to communicate with the CNMV on behalf of EquityCo and the Parties (together as joint bidders) to the extent they consider reasonably necessary to do so in relation to their relevant responsibilities in connection with the Offer. In addition, each of the Parties may communicate with the CNMV in relation to its own position under the Tender Offer Regulations provided that it shall promptly notify in writing Linklaters if any such communications with the CNMV take place in connection with the Transaction.

- **2.6** To the extent any public announcement or external communication is required by the CNMV or the Tender Offer Regulations to be made concerning the Offer on an urgent basis, the Parties shall use best endeavours to discuss the terms and contents of such public announcement or external communication and promptly make relevant representatives available on short notice for such purpose but, subject to the foregoing, such announcement or external communication may be made without the prior consent of the other Party.
- 2.7 Without prejudice to such withdrawing Party's obligations that otherwise subsist under this Agreement, if a Party has withdrawn from the Offer in accordance with Clause 9.1 it shall have no further obligations pursuant to Clauses 2.1 to 2.6 and 2.8 to 2.11 (inclusive) and its consent shall not be required for the purposes of such Clauses and Clauses 9.2.2, 12.4 and 13.1. Further, if (and for the duration of the period for which) a Party is in material breach of this Agreement or any Offer Documentation (including in respect of the Subscription of its Initial Investor(s) under the ECL (as defined therein) (an "Equity Commitment")), such

Party's consent shall not be required for the purpose of Clauses 2.1 to 2.6 and 2.8 to 2.11 (inclusive), 9.2.2, 12.4 and 13.1, and any director of the breaching Party appointed to the EquityCo Board or the board of any other JVCo shall not be entitled to attend any meeting of that board or vote on any resolution of that board.

- **2.8** Each Party undertakes to share with the other Party any material information available to it relating to Target or the Offer (including to the extent provided by any Adviser in connection with the Offer (other than such Advisers appointed solely to advise only one of the Parties)) from time to time, including information which is reasonably required:
 - (i) for public disclosure as required by the Tender Offer Regulations;
 - (ii) in connection with any regulatory filings required in any jurisdiction;
 - (iii) in connection with any process relating to the Avals (as defined in the ECL); or
 - (iv) in connection with the Financing, but excluding, in respect of any Party, any such material information which:
 - (a) relates to that Party only or which that Party is not permitted (under applicable law or regulation or any agreement with any third party) to share with the other Party; or
 - (b) is otherwise not permitted by (or conflicts with) Clause 12. Any commercially sensitive information (including Confidential Information) relating to either of the Parties required for any regulatory filings and/or approvals, shall be provided on a counsel to counsel basis.
- 2.9 The Parties agree that a bid committee (the "Offer Committee") shall be formed immediately following the execution of this Agreement. The Offer Committee shall be comprised of one representative appointed by each Party (the "Offer Committee Members") who shall, unless written notification (including by email) is provided of an alternative appointment, be:
 - **2.9.1** Linda Zhang as the TDR representative; and
 - 2.9.2 Alexander Metelkin as the ISQ representative.
- **2.10** All decisions of the Offer Committee require the unanimous approval of the Offer Committee Members other than as expressly provided in Clauses 2.7 (in case of a breach) and 6.4.3.
- **2.11** Each Party agrees that all consents, approvals, authorities or agreements required to be given by it pursuant to the terms of this Agreement (including the matters referred to in Clauses 2.3, 2.4, 9.2.2, 12.4 and 13.1) or otherwise in connection with the Offer shall be referred to the Offer Committee and shall be deemed to have been given by a Party if approved in writing by its Offer Committee Member.

3 Regulatory Filings

- **3.1** The Parties confirm they will continue to refine their analysis as to any anti-trust, or other regulatory filings or interactions which may be mandatory or advisable for the Offer and shall collaborate around obtaining necessary or appropriate information from Target.
- **3.2** The Parties shall work together in good faith with the intention of obtaining clarity between themselves on any mandatory or advisable filings and, in due course, on the process for making such filings.

4 Bid Financing

- **4.1** Subject to the Shareholders' Agreement, at the date on which the result of the Offer is published at the CNMV's website (or at such other date as is mutually agreed between the Parties in writing) (the "**Investment Date**"), in respect of the TDR Fund and the ISQ Funds (the "**Initial Investors**" and, together with any Permitted Syndicatee(s) to whom any commitment to fund any part of any Equity Commitment under the ECL (as defined below) is syndicated as part of any Permitted Syndication prior to the Investment Date in accordance with Clause 6.3, the "**Investors**"), respectively, shall commit (directly or indirectly) to EquityCo such amount of cash funding as set out in the ECL, such commitments to be made in the percentages set out below (the "**Initial Commitments**"):
 - 4.1.1 the TDR Fund: 50 per cent; and
 - 4.1.2 the ISQ Funds: 50 per cent,

and, subject to the terms of this Agreement, shall contribute some or all of such cash funding (the "**Initial Investments**") to subscribing (directly or indirectly) for securities or instruments in EquityCo (provided that the Initial Investor(s) of each Party will hold 50 per cent of the voting rights) pari passu in all respects and such Initial Investments will be made in securities or instruments of the same type and class, at the same price, and in the same proportions as between each such type and class. In connection with the Initial Commitments, the Initial Investors shall provide and be party to an equity commitment letter to EquityCo and the CCPs (as defined in the ECL) (the "ECL").

- **4.2** The Parties shall co-operate in good faith to ensure that EquityCo will have Financing in place in relation to the Offer as is required in order to comply with its obligations under the Tender Offer Regulations and that all cash funding is available as required to satisfy EquityCo's requirements in connection with the Offer.
- **4.3** If one or more of the Initial Investors (directly or indirectly) contributes cash funding to the JVCos pursuant to Clause 4.1 and in accordance with the ECL, and such cash funding is not used in full or the ECL subsequently terminates as a result of the expiry of the term of the ECL, the Parties shall procure (without limitation, but exercising such rights as they have in the relevant JVCo, and/or by instructing any director appointed to the relevant JVCo Board by that Party to vote in favour of any relevant resolution) that the relevant JVCo promptly returns such cash funding (or the cash funding surplus) to the Initial Investors in the amounts and proportions as contributed by or on behalf of each Initial Investor in accordance with the terms of the ECL.
- 4.4 If either Party fails to satisfy its obligations under the ECL (the "Defaulting Party") without prejudice to any other remedies that the other Party (as applicable) (the "Non-Defaulting Party") may have in respect of such failure:
 - **4.4.1** the Non-Defaulting Party may terminate this Agreement immediately upon giving written notice to the Defaulting Party;
 - **4.4.2** the Non-Defaulting Party may enforce the rights of EquityCo under the ECL, on behalf of EquityCo;
 - **4.4.3** the Defaulting Party shall, upon the Non-Defaulting Party's written election, immediately transfer, and shall procure that its Affiliates immediately transfer, to the Non-Defaulting Party, or as it may direct, any shares or other securities directly or indirectly held in EquityCo and/or any of the JVCos held by the Defaulting Party or

such Affiliate (provided, however, that if the Defaulting Party transfers shares or other securities directly or indirectly held in EquityCo and/or any of the JVCos to the Non-Defaulting Party pursuant to this provision, the Non-Defaulting Party shall refund the Defaulting Party any amounts previously funded by the Defaulting Party, subject to a reasonable right of set- off to cover Losses reasonably related to the default); and

4.4.4 the Defaulting Party shall indemnify the Non-Defaulting Party for any Losses incurred or suffered as a result of the Defaulting Party's failure to satisfy its obligations under the ECL, including Losses arising from any failure by EquityCo to implement the Offer resulting directly or indirectly from the Defaulting Party's failure to fund its Equity Commitment.

5 Standstill

- **5.1** Each Party confirms that, having made reasonable enquiry, neither it nor any of its Concert Parties is considered to hold any Target Securities Interests or have acquired Target Securities Interests in the 12 months prior to the date of this Agreement. For the avoidance of doubt the above representation and warranty is limited to the actual awareness of each Party of the Target Securities Interests of its Concert Parties as at the date of this Agreement.
- **5.2** Except pursuant to the Offer, from the date of this Agreement until the Relevant Date, neither Party shall and each Party shall procure that none of its Concert Parties (as from the date referred to in Clause 5.5) nor (so far as it is within its powers) any JVCo shall:
 - **5.2.1** either alone or acting in concert with others acquire or offer to acquire, or cause another person to acquire or to offer to acquire Target Securities Interests; or
 - **5.2.2** enter into an agreement or arrangement (whether conditional or otherwise) to do any of the matters set out in Clause 5.2.1.
- **5.3** Except for transfers to JVCos in connection with the Offer, neither Party shall, and each Party shall procure that none of its Concert Parties nor (so far as it is within its powers) any JVCo shall, from the date of this Agreement until the earlier of: (i) the date on which such Party withdraws from the Offer in accordance with Clause 9.1; and (ii) the Effective Date, sell, transfer or otherwise dispose of any Target Securities Interests or enter into an agreement or arrangement (whether conditional or otherwise) to do the same.
- **5.4** If the Parties announce that they are not proceeding with the Offer:
 - **5.4.1** the Parties (including, once incorporated, any JVCo) will each comply with the terms of any announcement made under the Tender Offer Regulations;
 - **5.4.2** the Parties will co-operate to satisfy the requirements that the CNMV may impose and
 - **5.4.3** neither Party shall (and each Party shall procure that none of its Concert Parties (including, so far as within its power, any JVCo) shall), either alone or acting in concert with others acquire or offer to acquire, any Target Securities Interests or enter into an agreement or arrangement as a result of which it or any person may acquire any Target Securities Interests, in each case only to the extent that: (i) the Parties are deemed to be Acting in Concert; or (ii) such action would result in the other Party or EquityCo being required to make a mandatory offer for Target.
- **5.5** Each Party agrees that it shall take all actions within its power to serve (or procure the service of) dealing stop notices to its Concert Parties as soon as practicable following the Filing Date

(to the extent that any Concert Party has not already been sent a dealing stop notice prior to that date).

6 Exclusivity and Syndication

- **6.1** Each Party warrants as at the date of this Agreement to the other that it is not a bidder, acquirer, lender to any such person, or otherwise an interested party in, any other bid or proposal in relation to the possible acquisition of some or all of the assets or share capital of Target and that it is not otherwise a part of, nor has agreed formally or informally to take part in or lend to, any form of partnership, joint venture, consortium or similar arrangement with/of any other party or parties making or contemplating making an offer for some or all of the assets or share capital of Target.
- 6.2 Each Party undertakes to the other Party, from the date of this Agreement:
 - 6.2.1 until the earlier of:
 - (i) the date on which such Party withdraws from the Offer in accordance with Clause 9.1; and
 - (ii) the date on which this Agreement terminates or is terminated in accordance with Clause 9.3,

to work with the other Party on an exclusive basis to further the Offer; and

6.2.2 until such Party's Relevant Date, to not, and to procure that none of its Affiliates nor its or their Representatives (excluding any lender, proposed lender, professional adviser, auditor or other representative not within such Party's control) or any of its Funds do not, except as part of the Offer, directly or indirectly be involved as an equity investor or as the provider of any other form of financing or otherwise perform any other substantive role or service (or enter into discussions or agree formally or informally to do the same) in respect of any acquisition of Target or any other transaction in relation to Target having a similar effect,

in each case, other than with the prior written consent of the other Party.

- 6.3 Without prejudice to any transfer rights agreed in the Shareholders' Agreement, each Party shall be entitled, from the date of this Agreement until the date of termination of this Agreement, to syndicate (or agree to syndicate) any of its Initial Investors' commitment to fund its Equity Commitment under the ECL and/or any indirect interest held by any Investor or Affiliate of any Investor in Amber JVCo securities, (as applicable, the "Syndication Interests"), to a Potential Syndicatee, in each case, without the consent of any other Party (a "Permitted Syndication", and such syndicatee, a "Permitted Syndicatee"), provided that:
 - **6.3.1** no syndication of (or agreement to syndicate) any indirect interest in Amber JVCo securities is permitted to any Restricted Syndicatee;
 - **6.3.2** such Party or an Affiliate of such Party retains all voting and other control rights (including all rights exercisable by such Party (or on its behalf) on the board of Amber JVCo or such other board as may control the Amber JVCo Group from time to time, or as an indirect shareholder in Amber JVCo) in connection with such Syndication Interests and such Syndication Interests are held by the Permitted Syndicatee indirectly in a fund or other vehicle managed or controlled by an Affiliate of such Party;

- 6.3.3 any Permitted Syndicatee to whom any Initial Investor syndicates any part of such Initial Investor's commitment to fund its Equity Commitment under the ECL shall make any contribution of cash pursuant to such commitment on the Investment Date to Amber JVCo indirectly and shall in no event acquire any direct interest in Amber JVCo securities;
- 6.3.4 the TDR Fund and its respective Affiliates maintain direct economic ownership of (and do not syndicate pursuant to any Permitted Syndication in respect of) at least 50.01 per cent. of its Initial Investment;
- 6.3.5 the ISQ Funds and their respective Affiliates maintain direct economic ownership of (and do not syndicate pursuant to any Permitted Syndication in respect of) at least 50.01 per cent. of its Initial Investment;
- **6.3.6** the relevant Party complies with the terms of Clause 6.4 (and procures the same in respect of any Potential Syndicatee); and
- **6.3.7** each relevant Initial Investor remains primarily responsible in respect of its full Equity Commitment given under the ECL at the Filing Date.
- 6.4 In connection with any Permitted Syndication:
 - **6.4.1** each Party may approach any person pursuant to Clause 6.3, provided that such person is a Potential Syndicatee;
 - **6.4.2** each Party shall be entitled to approach and discuss the Offer with any of their respective Potential Syndicatees, provided that:
 - (i) such Potential Syndicatee shall have executed:
 - (a) a non-disclosure agreement with customary provisions regarding the use of Confidential Information and in accordance with any applicable requirement of the Tender Offer Regulations and Market Abuse Regulation (596/2014); and
 - (b) applicable hold harmless letters prior to the disclosure to it of any reports prepared in connection with the Offer by advisers to the Parties;
 - (ii) any disclosure of Confidential Information to a Potential Syndicatee pursuant to this Clause 6 is made in accordance with the terms of Clause 12;
 - the Potential Syndicatee provides customary 'know your customer' and antimoney laundering information and documents relating to itself as reasonably requested by the Parties; and
 - (iv) the Parties comply with any relevant provisions of the Tender Offer Regulations applicable to any equity syndication in an Offer Period;
 - **6.4.3** following completion of any Permitted Syndication, each Permitted Syndicatee shall have information rights limited to receipt of:
 - the annual audited consolidated financial statements of the Amber JVCo Group, within 180 days after the accounting period to which they relate, and quarterly consolidated financial statements of the Amber JVCo Group, within 60 days after the quarter to which they relate; and

- (ii) other information in relation to the Amber JVCo Group to the extent approved by the Offer Committee or the EquityCo Board upon request by either Party (or its Offer Committee Member), acting reasonably and provided that the member of the EquityCo Board or Offer Committee Member of the TDR Fund or the ISQ Funds (as applicable) shall not be permitted to vote in respect of such request;
- **6.4.4** each Party shall keep the other Party reasonably informed of the progress in respect of any Permitted Syndication;
- 6.4.5 each Party shall carry out any Permitted Syndication in compliance with the Tender Offer Regulations, any rulings, requests or requirements of the CNMV and all other applicable laws and regulations (including the Market Abuse Regulation (596/2014), Companies Act UK, Companies Act Spain, the Financial Services and Markets Act 2000, the Financial Services Act 2012 and the Securities Market Act Spain); and
- 6.4.6 neither Party shall carry out any Permitted Syndication prior to the Effective Date, if such Permitted Syndication would require a material change to be made to, or supplement to be issued in respect of, the relevant documentation required to implement the Offer including, without limitation, a supplement to the Offer Prospectus.
- 6.5 Except in relation to any Permitted Syndication, each Party hereby undertakes that it will not, and shall procure that its Affiliates will not, prior to the Effective Date, directly or indirectly syndicate, assign or transfer (or agree to syndicate, assign or transfer) any right or interest granted to each such Party under this Agreement in relation to the Offer, including any interest in EquityCo and/or any other JVCo, to any actual or potential additional co-investor for it to participate in the Offer together with such Party or its Affiliates without the prior written consent of the other Party. Any such syndication, assignment or transfer (or agreement thereof) after the Effective Date shall be subject to the terms of the Shareholders' Agreement.

7 Warranties, Undertakings and Acknowledgements

- 7.1 Each Party warrants to the other Party that:
 - **7.1.1** it has the requisite power and authority to enter into this Agreement and there is no agreement, commitment or other understanding that:
 - would preclude or restrict such Party from entering into and performing its obligations under this Agreement or any agreement contemplated by this Agreement to be entered into by such Party, including the making of the Offer and consummation of a transaction if successful; or
 - (ii) would require any Party to allow any other person to elect to participate in the transactions contemplated by this Agreement;
 - **7.1.2** this Agreement when executed will constitute valid, binding and enforceable obligations of such Party;
 - **7.1.3** it has obtained the necessary internal approvals required to enter into this Agreement;
 - **7.1.4** it has not acquired Target Securities Interests in the 12 months prior to the date of this Agreement;

- **7.1.5** it is not in possession of material non-public information pursuant to the Market Abuse Regulation (596/2014); and
- **7.1.6** it has taken legal advice as to the implications of the Tender Offer Regulations as it applies to the Offer.
- 7.2 Each Party undertakes, in connection with the Offer, to:
 - **7.2.1** comply with, to procure that its Representatives and Affiliates under its control comply with, and to otherwise direct its Affiliates and Representatives not under its control to comply with:
 - (i) the Tender Offer Regulations and/or any rulings, requests or requirements of the CNMV; and
 - (ii) with all other applicable laws and regulations (including the Market Abuse Regulation (596/2014), Companies Act UK, Companies Act Spain, the Financial Services and Markets Act 2000, the Financial Services Act 2012 and the Securities Market Act Spain); and
 - **7.2.2** procure (so far as within its power) that each JVCo complies with the laws and regulations referred to in Clause 7.2.
- **7.3** Each Party acknowledges that notwithstanding any other provision of this Agreement, nothing in this Agreement shall require either Party to act or refrain from acting in a manner which would cause it or its Affiliates to be in breach of any applicable law or regulation or the Tender Offer Regulations.

8 Indemnification

- 8.1 Subject to Clause 8.2, in respect of any liability owed to any third party under any Indemnification Agreement (including any Losses suffered or incurred in connection with such liability) ("Indemnification Agreement Liability"), if one of the Parties (or its Affiliates (excluding members of the Amber JVCo Group)) (the "Payor") shall have paid to such third party/ies less than their Relevant Proportion of and the other Party (or its Affiliates (excluding members of the Amber JVCo Group)) (the "Payee") shall have paid to such third party/ies more than their Relevant Proportion, the Payor shall pay to the Payee such amount as shall be required so that after taking account of such payment the Payor and the Payee shall have borne their respective Relevant Proportions of the Indemnification Agreement Liability.
- **8.2** The Payee shall not be entitled to receive any amounts from the Payor pursuant to Clause 8.1 if the Indemnification Agreement Liability directly results from the fraud, wilful default or negligence of the Payee, or the breach by the Payee of its obligations to any such third party thereunder, in each case as determined by a court or arbitral tribunal of competent jurisdiction in a final and non-appealable order (or pursuant to a settlement agreement in relation thereto).
- **8.3** The Payor shall pay any amount payable pursuant to Clause 8.1 on or before the date falling 30 days after the same becomes due and payable.
- **8.4** All payments made by the Payor in respect of any Indemnification Agreement Liability shall be made in full without set-off or counterclaim whatsoever and without any tax deduction. If a tax deduction is required by law to be made by the Payor, the amount of the payment due from the Payor shall be increased to an amount which (after making any tax deduction)

leaves an amount equal to the payment which would have been due if no tax deduction had been required.

9 Withdrawal and Termination

- **9.1** Prior to the Filing Date, either Party may withdraw from the Offer and terminate this Agreement upon giving written notice to the other Party. Each Party will also notify the other Party promptly if its investment committee or equivalent approving body ceases to be supportive of the Offer, and the other Parties may deem such notice to constitute a notice of withdrawal from the Offer under this Clause 9.1.
- **9.2** If a Party withdraws from the Offer in accordance with Clause 9.1 above:
 - 9.2.1 such departing Party (the "Departing Party") shall cease to have any rights under this Agreement but shall remain subject to the applicable obligations set out in Clauses 1 (Definitions and Interpretation), 5 (Standstill), 6 (Exclusivity and Syndication), 10 (Transaction Costs), 13 (Assignment and Other Dealings), 14 (Amendments and Waivers), 15 (Severability) and 18 (Governing Law and Jurisdiction); and
 - **9.2.2** the remaining Party (the "**Remaining Party**") shall be entitled to progress and complete the Offer without the involvement of the Departing Party.
- **9.3** Save for the obligations set out in Clauses 1 (*Definitions and Interpretation*), 5 (*Standstill*), 6 (*Exclusivity and Syndication*), 7 (*Warranties, Undertakings and Acknowledgements*), 8 (*Indemnification*), 9 (*Withdrawal and Termination*), 10 (*Transaction Costs*), 13 (*Assignment and Other Dealings*), 14 (*Amendments and Waivers*), 15 (*Severability*) and 18 (*Governing Law and Jurisdiction*), and to the accrued rights of either Party, which shall each survive termination of this Agreement, the provisions of this Agreement shall terminate upon the earliest of the following to occur of:
 - **9.3.1** the Effective Date;
 - 9.3.2 the CNMV does not authorise the Offer;
 - **9.3.3** following the Filing Date, (i) the Parties agree, subject to Tender Offer Regulations, on the withdrawal of the Offer, or (ii) the Offer conditions are neither satisfied nor waived on time rendering the Offer ineffective;
 - **9.3.4** 21 September 2023, if the Offer Prospectus has not been filed with the CNMV by such date; and
 - **9.3.5** the Parties agreeing in writing to terminate this Agreement.

10 Transaction Costs

- **10.1** Joint Bid Costs, ISQ Costs and/or TDR Costs shall only be incurred in accordance with the Budget or otherwise with the prior agreement of the Parties and the Budget may not be amended without the prior approval of the Parties.
- **10.2** If the Offer is completed, the JVCos will pay all Joint Bid Costs in accordance with Clause 10.3 (as applicable). Each Party will, to the extent required, pay any Joint Bid Costs payable prior to the Effective Date pro rata in the Relevant Proportions by way of loan to any JVCo or such other method as the Parties may agree.

- **10.3** BidCo will pay any Joint Bid Costs related to the Transaction and FinCo will pay any Joint Bid Costs related to the Financing, in each case as contemplated by the Structure Paper.
- **10.4** Subject to Clause 11, notwithstanding the terms of any Engagement Letter between any Party or any JVCo, on one hand, and any adviser, on the other hand, and irrespective of the proportions in which any adviser seeks to recover or actually recovers from the Party or any one of them, if the Offer is not completed all Joint Bid Costs will be paid by each Party pro rata in the Relevant Proportions and, to the extent necessary, each Party shall take all such steps as may be necessary to give effect to such agreed proportionate sharing of liability (including, if required, contributing its share of Joint Bid Costs to any JVCo).
- **10.5** Any costs, fees and expenses incurred by a Party in connection with the Offer without the prior written approval of the other Party shall be for that Party's account.
- **10.6** Subject to Clause 10.7 below, if a Party becomes a Departing Party, the Departing Party will only be responsible for the Joint Bid Costs which are accrued up to the date of withdrawal (whether or not by then invoiced) pro rata to its Relevant Proportion (immediately prior to such withdrawal) or as otherwise agreed between the Parties. Any Departing Party will pay such share of the Joint Bid Costs on demand from time to time following presentation to such Party of each relevant invoice. Subject to the following sentence, a Departing Party shall not be responsible for any "tail fees" or similar of any advisers to or on behalf of any JVCo and/or any of the Parties providing financial advisory services which are incurred in connection with the Offer or otherwise. If any Party or their respective Affiliates (acting alone outside the scope of this Agreement) takes any action after the termination of the appointment of any such advisers which results in any "tail fees" or similar becoming payable to such adviser then such Party alone shall be liable in full for the payment of such fees.
- **10.7** If a Remaining Party (or any of its Affiliates) completes the acquisition of, or any similar transaction involving, Target or a controlling interest in its business, either alone or by jointly pursuing it with another person, then a Departing Party shall be reimbursed by such Remaining Party for all Joint Bid Costs previously paid by the Departing Party (in accordance with Clause 10.6) after its date of withdrawal in accordance with Clause 10.1, provided such acquisition is completed within 12 months of the Departing Party's withdrawal.

11 Engagement and Reports and Reliance

- 11.1 Each Party acknowledges that, as at the date of this Agreement, either Party or one of their Affiliates have entered into Engagement Letters and have the benefit of all Reports and Reliance Letters. On or prior to the Effective Date, each Party shall use reasonable endeavours to procure that the Engagement Letters and any rights it or any of its Affiliates may have pursuant to the Report and Reliance Letters are novated and extended to, with effect from not later than immediately prior to the Effective Date, to EquityCo and/or any of the other JVCos, such that on or following the Effective Date, EquityCo and/or any of the other JVCos shall be entitled to make a claim against each adviser under an Engagement Letter or a Report and Reliance Letter and any payment of fees and expenses shall either be made by: (i) EquityCo and/or any of the other JVCos).
- **11.2** To the extent the rights and obligations of either Party or one of their Affiliates pursuant to an Engagement Letter or a Report and Reliance Letter are not novated to EquityCo and/or any of the other JVCos, until such time as a particular Engagement Letter or a Report and Reliance Letter is so novated:

- 11.2.1 each Party shall, and shall procure any of its Affiliates shall, take all such action as is reasonably requested by EquityCo, any of the other JVCos and/or the other Party to make recovery under such Engagement Letter or a Report and Reliance Letter and shall account to EquityCo and/or any of the other JVCos for any amounts so recovered (and shall hold such amounts on trust for EquityCo and/or any of the other JVCos) of any claim or right in respect of such Engagement Letter or Report and Reliance Letter; and
- **11.2.2** EquityCo and/or any of the other JVCos shall indemnify and hold harmless (on a continuing basis) each Party and each of its Affiliates that is a party to an Engagement Letter from and against (and to pay on demand an amount equal to) any and all Losses incurred or suffered in connection with such Engagement Letter, provided that such Party acts in accordance with the terms of the Engagement Letter.

12 Confidentiality and Announcements

- **12.1** Each Party shall, and shall procure that its Affiliates to whom it discloses Confidential Information shall, treat as strictly confidential all Confidential Information.
- **12.2** Each Party acknowledges that a copy of this Agreement will be disclosed as part of the information included in the Offer Prospectus.
- 12.3 Each Party acknowledges, and shall inform its Affiliates to whom it discloses Confidential Information, that some or all of the Confidential Information may be information which is not public or otherwise generally available and is of a kind such that a person who has that information would be prohibited or restricted from using it to deal in the securities of Target under the Market Abuse Regulation, Part V Criminal Justice Act 1993, Securities Market Act Spain, the Criminal Code Spain or other applicable insider dealing, market abuse or similar laws.
- **12.4** Subject to Clauses 12.5 to 12.6, each Party hereby undertakes to the other Party that it shall not, and shall procure that its Affiliates to whom it discloses Confidential Information shall not, except with the prior written consent of the Parties, make use of (save for performing its obligations under this Agreement) or disclose to any person any Confidential Information.
- **12.5** Each Party may disclose Confidential Information:
 - **12.5.1** to its Representatives and its Affiliates;
 - **12.5.2** to any professional adviser to the extent that such person strictly needs access to that Confidential Information for the purpose of evaluating, negotiating, advising upon or implementing the Offer, provided that such Party:
 - (i) informs such person that the Confidential Information is confidential and of the existence and terms of this Clause 12; and
 - (ii) procures that any such person complies with the provisions of this Clause 12 as if they were a party to them; and
 - **12.5.3** if and to the extent required by:
 - any order, direction or ruling of any court of competent jurisdiction, governmental, regulatory or supervisory body;
 - (ii) the laws or regulation of any country with jurisdiction over the affairs of the relevant Party; or

- (iii) any securities exchange, listing authority or any regulatory or supervisory body (including the CNMV) provided that (to the extent permitted by law) it consults with the other Party as to the contents of such disclosure and, in any event, discloses only the minimum information necessary in order to satisfy such requirement.
- **12.6** Clauses 12.1 and 12.4 will not apply to Confidential Information which:
 - **12.6.1** at the time of supply is in the public domain;
 - **12.6.2** subsequently comes into the public domain otherwise than as a result of the breach of this Agreement;
 - **12.6.3** a Party can establish to the reasonable satisfaction of the other Party, is independently developed by such Party, its Affiliates, Representatives or professional advisers; or
 - **12.6.4** a Party can establish to the reasonable satisfaction of the other Party that it is already in its lawful possession or that of any of its Affiliates or professional advisers and is free from any obligation of secrecy or confidence or it subsequently comes lawfully into a Party's possession or that of any of its Affiliates or professional advisers from a third party source which source does not have any obligation of confidentiality in relation to such Confidential Information.

13 Assignment and Other Dealings

- **13.1** This Agreement is personal to the Parties and neither Party shall assign, transfer, mortgage, charge, subcontract, declare a trust over or deal in any other manner with any of its rights and obligations under this Agreement.
- **13.2** Nothing in this Agreement is intended to, or shall be deemed to, establish any partnership between either Party, constitute either Party as the agent of the other, or authorise either Party to make or enter into any commitments for or on behalf of any other.

14 Amendments and Waivers

- **14.1** No amendment to, or waiver of any of the provisions of, this Agreement shall be effective unless in writing and signed by or on behalf of each of the Parties.
- **14.2** No delay or omission by any party in exercising any right or remedy provided by law or under this Agreement shall constitute a waiver of such right or remedy.

15 Severability

If any provision in this Agreement shall be held to be illegal, invalid or unenforceable, in whole or in part, under any enactment or rule of law or otherwise, such provision (or part) shall to that extent be deemed not to form part of this Agreement but the legality, validity and enforceability of the remainder of this Agreement shall not be affected.

16 Counterparts

This Agreement may be executed in any number of counterparts and by the Parties to it on separate counterparts and each such counterpart shall constitute an original of this

Agreement but all of which together constitute one and the same instrument. This Agreement shall not be effective until each Party has executed at least one counterpart.

17 Third party rights

The Contracts (Rights of Third Parties) Act 1999 shall not apply to this Agreement and no person who is not a party to this Agreement may enforce any provision of it.

18 Governing Law and Jurisdiction

- **18.1** This Agreement and any non-contractual obligations arising out of or in connection with it (including any non-contractual obligations arising out of the negotiation of the transaction contemplated by this Agreement) are governed by and shall be construed in accordance with the laws of England and Wales.
- **18.2** The Parties irrevocably agree that the courts of England shall have exclusive jurisdiction to settle any dispute or claim that arises out of or in connection with this Agreement (including a dispute relating to any non-contractual obligation arising out of or in connection with either this Agreement or the negotiation of the transaction contemplated by this Agreement).

Schedule 1 Parties

Part A – TDR Fund

 TDR Capital V L.P., an English limited partnership, as constituted from time to time, with registered number LP022040 having its principal place of business at 20 Bentinck Street, London W1U 2EU

Part B – ISQ Funds

- 1. ISQ Global Infrastructure Fund III (UST), LP, a Cayman Islands limited partnership, whose registered office is at c/o Maples Corporate Services Limited, PO Box 309, Ugland House, Grand Cayman, KY1-1104, Cayman Islands
- ISQ Global Infrastructure Fund III (USTE), LP, a Cayman Islands limited partnership, whose registered office is at c/o Maples Corporate Services Limited, PO Box 309, Ugland House, Grand Cayman, KY1-1104, Cayman Islands
- **3.** ISQ Global Infrastructure Fund III, LP, a Cayman Islands limited partnership, whose registered office is at c/o Maples Corporate Services Limited, PO Box 309, Ugland House, Grand Cayman, KY1-1104, Cayman Islands
- **4.** ISQ Global Infrastructure Fund III (EU), LP, a special limited partnership (*société en commandite spéciale*), whose registered office is at 6, rue Eugène Ruppert, L-2453 Luxembourg, Grand Duchy of Luxembourg

Schedule 2 Term Sheet

Project Amber

Term Sheet

This non-binding term sheet summarises the principal terms on which TDR Capital General Partner V Limited acting in its capacity as general partner of the TDR Fund ("TDR") and ISQ Global Fund III GP, LLC acting in its capacity as general partner of various ISQ Funds ("ISQ") (each an "Investor" and together the "Investors") propose to hold interests in Amber JVCo in connection with the proposed acquisition of the company code-named "Amber" (the "Transaction"), and which the Investors have agreed to follow in the negotiation and conclusion of a binding shareholders' agreement to govern the arrangements relating to their interests in Amber JVCo (and together with its subsidiary undertakings, the "Amber JVCo Group") (the "Shareholders' Agreement").

Terms used but not defined in this document shall have the meanings given to such terms in the joint bid agreement to be entered into between TDR and ISQ in relation to the Transaction (the "**Joint Bid Agreement**").

Provision	Description of Terms
General Structure	Amber JVCo will be an English private limited company tax resident in England.
	The structuring of the Transaction and nature of the instruments to be subscribed by the Investors will be determined in accordance with tax and operational advice to be obtained by the Investors.
	Subject to legal and tax advice, each Investor will fund their Equity Commitment by way of a subscription for the same class and type of securities in Amber JVCo.
ISQ Holding Structure	ISQ will hold its interest in Amber JVCo through two entities: one incorporated in the United Kingdom (" ISQ UK Shareholder ") and the other incorporated in the Cayman Islands (" ISQ CI Shareholder ").
	Any provision which is expressed to bind the ISQ UK Shareholder and the ISQ CI Shareholder shall, save where inconsistent with the context, bind each of them severally and not jointly and severally.
	All rights and obligations of the ISQ CI Shareholder shall be exercised by the ISQ UK Shareholder acting on behalf of the ISQ CI Shareholder. For the avoidance of doubt, this includes any consents required to be given by the ISQ CI Shareholder.
Further Funding	There shall be no obligation on the Investors to provide further funding to the Amber JVCo Group after Completion.
Board Composition	The Amber JVCo board (the "Amber JVCo Board") will consist of at least 8 directors.

Provision	Description of Terms
	Each Investor shall have the right to appoint and remove up to four directors (each, an " Investor Director ") or such other greater number of directors as agreed between the Investors (provided that each Investor shall be entitled to four board votes, irrespective of the number of Investor Directors they have appointed).
	The Investors also expect that the Amber JVCo Group will establish a board which will be responsible for the operation and management of Amber (the " Operational Board "). The Operational Board is expected to be established post-Completion following discussion with the CEO of Amber and is expected to comprise of senior management of Amber and up to four Investor Directors for each Investor. Voting and quorum of the Operational Board will be as per the Amber JVCo Board (see below), save that certain matters may be reserved to the Amber JVCo Board and the positive vote of each Investor is required for any matter to be resolved at the Operational Board.
	Amber JVCo Board meetings will only be quorate if at least one Investor Director representing each Investor is present.
	Amber JVCo Board resolutions will be passed by a simple majority of the Amber JVCo Board (with each Investor having a total of four votes), except for resolutions with respect to the Reserved Matters (as defined below), for which any resolution to be passed by the Amber JVCo Board must include the approval of at least one Investor Director appointed by each of the Investors. Any Investor Director shall be entitled to exercise the votes of all Investor Directors appointed by their respective Investor.
	Each Investor shall have the right to have equal representation on each committee, and the boards or committees of any subsidiary.
	The Shareholders' Agreement will contain customary sunset provisions which reduce the number of Investor Directors which an Investor can appoint if the equity interest in Amber JVCo held by such Investor is reduced.
Delegation of Authority	The Investors shall agree a policy for the delegation of authority from the Amber JVCo Board to the management team of Amber for the operational day-to-day management of the business, subject always to the Reserved Matters (as defined below).
Reserved Matters	The Shareholders' Agreement shall include a customary list of "reserved matters" in respect of the Amber JVCo Group (including those set out in Appendix 1) that will require the prior written consent of each of the Investors (the " Reserved Matters ").
	The Shareholders' Agreement will contain customary sunset provisions which reduce the number of Reserved Matters which will require the prior written consent of each of the Investors if the equity interest in Amber JVCo held by such Investor is reduced.

Provision	Description of Terms
Positive Covenants	The Shareholders' Agreement shall include a customary list of positive covenants (including those set forth in Appendix 2) that each Amber JVCo Group company shall be required to comply with and management shall be required to procure compliance with such covenants.
Deadlock	A deadlock will arise if either:
	(i) the Amber JVCo Board does not pass a resolution put to it two or more times; or
	(ii) two or more consecutive Amber JVCo Board meetings have been adjourned because a quorum is not present.
	If a deadlock arises, the relevant matter shall be escalated to Gary Lindsay in the case of TDR and Mohamed El Gazzar in the case of ISQ.
	If following escalation, the Investors they cannot come to a mutually satisfactory resolution within 10 business days of the matter being put to them, the Amber JVCo Group's business will carry on in the ordinary course and the <i>status quo ante</i> shall prevail.
	For the avoidance of doubt, these deadlock provisions shall only apply to disagreements as between the Investors.
Information rights	Amber JVCo will provide each Investor with:
	(i) annual audited accounts;
	(ii) six-monthly reporting (to include a half-year forecast and presentation from the Amber JVCo Board);
	(iii) quarterly reporting (to include a quarterly report from Amber JVCo);
	(iv) monthly reporting;
	(v) any updates to, or deviations from, the business plan;
	(vi) all materials provided to any Amber JVCo Board;
	(vii) details and copies of all material interactions with regulators regarding the Amber JVCo Group or issues material to the Amber JVCo Group;
	(viii) any information disclosed to the lenders under the Financing Documents (as defined below);
	(ix) any information required by either Investor in relation to environmental, social or governance matters; and
	(x) other information (to include minutes of any Amber JVCo Board meetings, as well as any other information reasonably requested by an Investor from time to time).

Provision	Description of Terms
Anti-Dilution / Pre- emption Rights	An issue of new equity or debt securities in any Amber JVCo Group company will be a Reserved Matter (as defined below) (unless a default event has occurred).
	Each Investor will have the right to participate on an issue of new securities in cash pro rata to their holding of shares, with customary exceptions (e.g. intra-group, M&A consideration, management incentive plans etc.).
Default Event	If there is either a prospective or an actual event of default under the Amber JVCo Group's third party financing documents (the "Financing Documents") or another immediate cash need to solve for a credit issue in respect of the Amber JVCo Group, a default event shall be deemed to have occurred.
	Following a default event, either Investor shall be entitled to require that new equity is issued without the consent of the other. Any such issue shall provide for a 20 business day catch period for the other party and shall otherwise comply with the procedures for the issue of new equity.
	Any new equity must be issued at fair market value, as agreed between the Investors or, failing agreement, as determined by an independent accountant.
Transfer Restrictions	The transfer restrictions below (including under right of first offer, drag and tag) shall be subject to customary "permitted transfer" exceptions (including for the avoidance of doubt transfers to affiliates).
	Lock-Up Period
	Neither Investor shall be permitted to dispose of any equity or debt interests in the Amber JVCo Group without the consent of the other for two years from Completion (the "Lock-Up Period").
	Following Lock-Up Period
	All sales by an Investor shall be subject to ROFO, drag and tag as set out below.
Right of First Offer ("ROFO")	Prior to any sale process for any number, whether in whole or in part, of the shares held by an Investor (the " Sale Shares "), such selling Investor shall inform the other Investor in writing and specify the price per share at which they are willing to sell (the " ROFO Price ").
	The non-selling Investor shall have two months to submit a binding fully financed offer for all of the selling Investor's Sale Shares (the " ROFO Offer ").
	If the ROFO Offer is for a price greater than or equal to the ROFO Price, the selling Investor shall be required sell its Sale Shares to the non-selling Investor.

Provision	Description of Terms
	If the ROFO is for a price less than the ROFO Price, the selling Investor can either accept this offer or launch a market exercise, in which the non-selling Investor may also participate, join in any discussions regarding the process and other bidders and will have the right to be kept informed of progress. If the market exercise does not result in a price greater than or equal to the ROFO Price, the selling Investor cannot sell its Sale Shares for a period of twelve months following the conclusion of the failed ROFO process.
Drag Along Rights	Following the second anniversary of Completion, subject to completion of the ROFO process, an Investor which is selling all of its interests in the Amber JVCo Group to a bona fide third party can drag the other Investor, provided that the consideration received by the dragged Investor would represent a multiple of money invested for the dragged Investor of at least 3.0x. For the avoidance of doubt, ISQ may only exercise the drag along rights described above if both the ISQ UK Shareholder and the ISQ CI Shareholder are selling their respective interests in the Amber JVCo Group.
	The terms of the drag-along sale shall be the same price per security and on the same terms (including as to type of consideration), subject to the following.
	The Investor being dragged shall (i) receive cash or liquid securities as consideration for its Amber JVCo Group securities, and (ii) be required to give the same indemnities, the same warranties, the same non-solicitation covenants (but not non-compete covenants) and place the same amount of proceeds in escrow as the dragging Investor.
Tag Along Rights	The non-selling Investor shall have pro rata tag along rights on any sale of shares by the selling Investor.
Drag and Tag Costs	The Shareholders' Agreement will contain customary provisions relating to the treatment of any costs incurred by Amber JVCo/the Investors in connection with any drag-along/tag-along sales.
IPO	Same treatment as sales of shares in that not permissible during the Lock-Up Period unless the Investors otherwise agree. Following the fifth anniversary of Completion, either Investor can initiate an IPO. No Investor shall be obliged to sell down as part of the IPO. Each Investor to have the option to tag along and sell secondary shares in the IPO in proportion to its respective shareholding. Customary IPO co-operation and lock up language to be included. Each Investor to have equivalent registration rights in respect of the IPO.

Provision	Description of Terms
Notification / Information Rights on Exit	All approaches to be disclosed ASAP to the Amber JVCo Board/the Investors. Any appointment of advisers to be a Reserved Matter. Investors to have rights to receive any and all information provided to potential purchasers. Disclosure of all customary financial, technical and legal information to potential bidders permitted by the Amber JVCo Group and/or either Investor, provided that the recipients have entered into confidentiality undertakings for the benefit of the Amber JVCo Group in a customary form.
Management incentive plan	The Investors shall, between them, agree on the terms and the structure of the management incentive plan following Completion, which will dilute each Investor on a <i>pro rata</i> basis.
Costs, Fees and Expenses	Amber JVCo to pay (or procure that another Amber JVCo Group company pays) certain costs, fees and expenses to each Investor on a <i>pro rata</i> basis (in respect of fees only and not costs or expenses) and subject to an overall cap to be agreed between the Investors.
Other	The governing law of the transaction documents will be English law. Any costs to be borne by the Amber JVCo Group (including each Investor's costs in relation to negotiating the consortium arrangements) will be agreed in accordance with the Joint Bid Agreement.

Appendix 1 Reserved Matters

No Amber JVCo Group company shall take any of the following actions without the prior written consent of each of the Investors:

1.	Alter its constitutional documents.
2.	Change the name of any Amber JVCo Group company.
3.	Adopt its annual budget and/or business plan, or vary the existing budget/business plan, or exceed the capex provided for in any business plan by more than a percentage threshold to be agreed between the Investors.
4.	Allot or issue any shares or other debt or equity securities or grant to any person any option or right to call for the issue of any shares or other securities (save to another Amber JVCo Group company) other than in accordance with the Shareholders' Agreement or articles, repurchase or redeem any debt or equity securities, or effect any other variation to its issued share capital, share premium account, or its capital structure, or the rights attaching to any equity or debt securities or capitalize any reserves or reduced the amount standing to the credit of any reserve.
5.	Recommend, declare or pay a dividend or other distribution to any person other than another Amber JVCo Group company.
6.	Create or issue or allow to come into being any security interest (other than a lien on assets arising by operation of law in the ordinary course of business and securing sums not more than 30 days overdue or as envisaged by the Financing Documents) over any part of its property or assets or create or issue any debenture or debenture stock.
7.	Appoint or remove (other than as an alternate pursuant to the relevant group company's articles) a person as a director of an Amber JVCo Group company (other than in accordance with the terms of the Shareholders' Agreement and the articles of the relevant Amber JVCo Group company).
8.	Appoint (except for the reappointment of its existing auditors) or remove its auditors, adopt any new accounting policy, make any material changes to any material accounting policies or change its accounting reference date, in each case save as required to comply with law or a new accounting standard.
9.	Adopt the audited accounts of Amber JVCo and the audited consolidated accounts of the Amber JVCo Group.
10.	Acquire an interest (whether on its own behalf or as a nominee) in the share, loan capital or instruments convertible into the share capital of any company or other legal entity or its business or assets or enter into any formal discussions or negotiations in connection therewith.

11.	Whether by a single transaction or by a series of transactions: (a) acquire, sell, transfer or enter into an agreement for the acquisition, sale, transfer, surrender or other disposition of any assets having a book or market value in excess of a \in threshold to be agreed between the Investors or acquire or enter into an agreement for the acquisition of any assets having a book or market value in excess of a \in threshold to be agreed between the Investors or acquire or enter into an agreement for the acquisition of any assets having a book or market value in excess of a \in threshold to be agreed between the Investors (except, in each case, where already identified and agreed as part of the budget sign off from time to time); or (b) enter into, materially vary or terminate any lease, licence, tenancy or similar arrangement where the rental and all other payments under it exceeds a \in threshold per annum to be agreed between the Investors.
12.	Acquire or dispose of any freehold or leasehold property, grant or surrender a lease in respect of such property or take or omit to take any action which could prejudice the continuation of any such lease.
13.	Make a material variation to, or waive a condition of, the Financing Documents, or voluntarily pre-pay any sums lent under the Financing Documents or refinance any such indebtedness, or enter into formal discussions or negotiations in connection thereto.
14.	Except for any debt facilities entered into in connection with the Offer: (a) borrow any money or obtain credit (other than normal trade credit); (b) make any other arrangement having a similar effect (including, without limitation, debt factoring, invoice discounting, hire purchase, equipment leasing, conditional or credit sales, or any off balance sheet borrowings); or (c) materially vary the terms of any credit arrangement, in each case, with any person other than an Amber JVCo Group company, if the aggregate amount outstanding from time to time (including sums attributable to capital under the then current accounting practice) exceeds a € threshold to be agreed between the Investors.
15.	Make, increase or extend a loan or advance exceeding a € threshold to be agreed between the Investors in aggregate to any person (including any loan and advance to a person connected with that person) (excluding (i) any other Amber JVCo Group company, (ii) trade credit in the ordinary course of trading, or (iii) advances made to employees against expenses properly incurred by them on an Amber JVCo Group company's behalf), or acquire any indebtedness owed by any Amber JVCo Group company or other third party to any lender.
16.	Give, extend or increase any liability a guarantee or indemnity, other than as required pursuant to the Financing Documents or otherwise in the ordinary course of business.
17.	Save in accordance with the Shareholders' Agreement, approve or register the transfer (whether legal or beneficial) of any shares in its capital or the price at which such transfer occurs, sell, liquidate or otherwise dispose of any subsidiary or any interest therein, or all or part of its assets, amalgamate, merge or demerge with any company or concern or otherwise effect any corporate restructuring or group reorganisation.
18.	Carry on part of an Amber JVCo Group company's business other than through an Amber JVCo Group company or become or cease to be party to, or vary materially the terms of participation in, a partnership, joint venture, consortium, or any other incorporated or unincorporated association (except for trade associations and existing partnerships and joint ventures of Amber or its subsidiaries in place as at Completion).

19.	Enter into, modify, waive, fail to enforce or terminate (or give notice to terminate or commit a material or persistent breach of) any material contract (which shall include contracts which: (i) have a net impact on Amber JVCo's consolidated EBITDA of a € threshold to be agreed between the Investors / will involve a payment or commitment by the Amber JVCo Group in excess of a € threshold to be agreed between the Investors or more; (ii) are outside the ordinary course of the Amber JVCo Group's business; (iii) is not terminable by the Amber JVCo Group on less than 12 months' notice; or (iv) is otherwise material).
20.	Commence, discontinue or settle any dispute, litigation or arbitration proceedings where the amount claimed (either by or against it) together with any costs incurred (or likely to be incurred) by it is in excess of a € threshold (exclusive of VAT) to be agreed between the Investors.
21.	Either (i) propose any resolution to place itself or any other Amber JVCo Group company in voluntary liquidation or administration or receivership or relating to a composition with its creditors generally, (ii) make a proposal for a voluntary arrangement under section 1 of the Insolvency Act 1986 in respect of any Amber JVCo Group company, or obtain a compromise or arrangement under Part 26 of the Act in respect of any Amber JVCo Group company or (iii) do anything similar or analogous to the matters described in (i) or (ii) above.
22.	Adopt, or materially amend the terms of, any management incentive plan or other bonus, profit sharing, share option or other incentive scheme (including related issues or grants of equity and equity-linked securities).
23.	Enter into, materially vary the terms of, terminate or give a board or other consent or approval in relation to, a related party transaction with a senior manager or an Investor and/or its affiliates.
24.	Appointment, dismissal, terms of employment (and any material variation thereto) and remuneration (including bonus and other incentive payments) of the Amber JVCo Group CEO, COO, CFO and other key management.
25.	Material changes to the nature, geographical area, and strategy of any Amber JVCo Group company's business, including carrying on a new business.
26.	Engage any professional advisers (i) with aggregate annual fees in excess of a € threshold to be agreed between the Investors (other than advisers in relation to matters within the ordinary course of its business) or (ii) in connection with an Exit.
27.	Apply for, allow to lapse or materially amend a regulatory approval or licence in any jurisdiction (other than renewals of any approvals or licences in the ordinary course of business).
28.	Make any material alteration to any insurance policy held by any Amber JVCo Group company.
29.	Grant or enter into any licence, agreement or arrangement concerning any part of the name or trading names of any Amber JVCo Group company or the goodwill attaching to the same or any other part of an Amber JVCo Group company's intellectual property.

30.	Make any political or charitable contribution or any other gift of whatsoever nature.
31.	Make an announcement in relation to any of the actions described in this Appendix 1 or in relation to a proposal to take any such action.
32.	Approve or agree to do any of the matters listed in this Appendix 1.

Appendix 2 Conduct of Business – Positive Covenants

Each Amber JVCo Group company shall:

1.	Procure that:
	 (a) any expansion, development or evolution of the business of the Amber JVCo Group, as carried on as at the date of the Shareholders' Agreement, is effected only through Amber JVCo Group companies;
	(b) the business of the Amber JVCo Group is conducted responsibly in accordance with principles of good corporate governance, best practice and high ethical standards and with due consideration to the reputation of the Amber JVCo Group and the Investors;
	(c) introduce and maintain effective and appropriate control systems in relation to the financial, accounting and record-keeping functions of the Amber JVCo Group; and
	(d) it promptly adopts and implements any corporate social responsibility, environmental or anti-corruption policies which either Investor may require from time to time.
2.	Keep in force and maintain at all times the policies referred to below or such other policies as may be acceptable to the Investors in substitution for them and will not take or omit to take any action or permit any action to be taken which might invalidate any such policy:
	(a) full and proper directors' and officers' liability insurance on terms reasonably acceptable to the Investors in respect of such persons (including any Investor Director) as the Investors may require and which shall provide for a minimum cover of a € threshold to be agreed between the Investors; and
	(b) full and proper insurance against such business risks and liabilities as the Investors may require (including key man insurance policies in respect of specific managers) with an insurance company approved by Investor consent on such terms and in such amounts as shall accord with good commercial practice (or as may otherwise be required from time to time by an Investor) and each Amber JVCo Group company shall procure that such insurances are reviewed by a reputable insurance broker at least once in each calendar year and that all reasonable recommendations of such broker are complied with.
3.	Maintain all licences, consents and authorisations whatsoever which are required or necessary to carry on the business of each Amber JVCo Group company from time to time.
4.	Comply with the requirements of any relevant regulator from time to time.
5.	Hold meetings of the board of directors in compliance with the terms of the Shareholders' Agreement.

6.	Procure that each of the other Amber JVCo Group companies shall pay to it (or, as the case may be, its immediate holding company) by way of dividend or management charge such sum as such Amber JVCo Group company may lawfully pay and as shall be required to permit the payment by the relevant Amber JVCo Group companies on the relevant date of any dividends payable on shares and of any amount payable on the redemption of any securities, subject to the Financing Documents and Investor consent.
7.	Act in accordance with any Investor directions in connection with the enforcement of its rights under any of the transaction documents or the Financing Documents.
8.	Maintain appropriate policies and procedures to assess anti-corruption, anti-competition, anti-money laundering and sanctions risk and to ensure compliance with these and any other compliance measures, and report to the Amber JVCo Board annually in relation to the Amber JVCo Group's compliance with these policies and procedures.
9.	Take all steps required by an Investor (or its Affiliates) for the Amber JVCo Group, the Investors and/or their Affiliates to comply with their obligations in relation to compliance measures, on a timely basis, including the payment or reimbursement of any costs for which the Investors and/or their Affiliates become liable.
10.	Procure that, other than with specific Investor consent, transaction, arrangement or agreement between (i) any Amber JVCo Group company and (ii) the Investors or any of its Affiliates, shall be on an arms-length basis.

THIS AGREEMENT HAS BEEN ENTERED INTO ON THE DATE STATED AT THE BEGINNING OF IT.

.....

duly authorised for and on behalf of

TDR Capital V L.P.

By: TDR Capital General Partner V L.P., its general partner

By: TDR Capital General Partner V Limited, its general partner

Name:

duly authorised for and on behalf of

ISQ Global Infrastructure Fund III, L.P.

By: ISQ Global Fund III GP, LLC, its general partner

By: ISQ Holdings, LLC, its sole member

Name:

duly authorised for and on behalf of

ISQ Global Infrastructure Fund III (UST), L.P.

By: ISQ Global Fund III GP, LLC, its general partner

By: ISQ Holdings, LLC, its sole member

Name:

duly authorised for and on behalf of

ISQ Global Infrastructure Fund III (USTE), L.P.

By: ISQ Global Fund III GP, LLC, its general partner

By: ISQ Holdings, LLC, its sole member

Name:

duly authorised for and on behalf of

ISQ Global Infrastructure Fund III (EU), L.P.

By: I Squared Capital Advisors (US) LLC, acting in its capacity as alternative investment fund manager

Name: